RESEARCH ARTICLE



REVISED The person-based development and realist evaluation

of a pre-consultation form for GP consultations [version 2;

peer review: 2 approved]

Mairead Murphy¹, Chris Salisbury¹, Anne Scott¹, Lucia Sollazzi-Davies², Geoff Wong³

¹Bristol Medical School, University of Bristol, Bristol, BS8 2PS, UK
 ²West Walk GP Practice, Bristol, UK
 ³Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, OX2 6HT, UK

V2 First published: 17 Feb 2022, 2:19 https://doi.org/10.3310/nihropenres.13249.1 Latest published: 29 Jul 2022, 2:19 https://doi.org/10.3310/nihropenres.13249.2

Abstract

Background: Use of telephone, video and e-consultations is increasing. These can make consultations more transactional, potentially missing patients' concerns. This study aimed to develop a complex intervention to address patients' concerns more comprehensively in general practice and test the feasibility of this in a cluster-randomised framework.

The complex intervention used two technologies: a patient-completed pre-consultation form used at consultation opening and a doctorprovided summary report provided at consultation closure. This paper reports on the development and realist evaluation of the preconsultation questionnaire.

Methods: A person-based approach was used to develop the preconsultation form. An online questionnaire system was designed to allow patient self-completion of a form which could be shared with GPs. This was tested with 45 patients in three rounds, with iterative adjustments made based on feedback after each round.

Subsequently, an intervention incorporating the pre-consultation form with the summary report was then tested in a cluster-randomised framework with 30 patients per practice in six practices: four randomised to intervention, and two to control. An embedded realist evaluation was carried out. The main feasibility study results are reported elsewhere.

Results:

Intervention Development: 15 patients were recruited per practice. Twelve patients, six GPs and three administrators were interviewed and 32 changes were made iteratively in three rounds. Recruitment

Open Peer Review Approval Status 1 2 version 2 (revision) 29 Jul 2022 version 1 17 Feb 2022 view view view view

 Seán Paul Teeling ^D, University College Dublin (UCD), Dublin, Ireland Queen Margaret University, Edinburgh, UK

2. Abi Eccles (D), University of Warwick, Coventry, UK

Any reports and responses or comments on the article can be found at the end of the article.

rates (proportion of patients responding to the text) increased from 15% in round one to 50% in round three.

Realist evaluation: The pre-consultation form was most useful for people comfortable with technology and with hidden concerns or anxiety about the consultation. It resulted in more issues being discussed and support provided, more effective use of time and greater patient satisfaction.

Conclusions: The person-based approach was successful. The preconsultation form uncovers more depth and improves satisfaction in certain consultations and patients. Technological improvements are required before this could be rolled out more widely.

Keywords

e-PROs, GP consultations, GP-patient communication, person-based approach, realist evaluation

Corresponding author: Mairead Murphy (mairead.murphy@bristol.ac.uk)

Author roles: Murphy M: Conceptualization, Data Curation, Formal Analysis, Funding Acquisition, Investigation, Methodology, Project Administration, Resources, Software, Supervision, Validation, Visualization, Writing – Original Draft Preparation; Salisbury C: Conceptualization, Funding Acquisition, Investigation, Methodology, Validation, Writing – Review & Editing; Scott A: Data Curation, Formal Analysis, Investigation, Project Administration, Validation, Writing – Review & Editing; Sollazzi-Davies L: Methodology, Resources, Writing – Review & Editing; Wong G: Conceptualization, Formal Analysis, Funding Acquisition, Investigation, Methodology, Validation, Writing – Review & Editing;

Competing interests: Geoff Wong is Deputy Chair of the United Kingdom's National Institute of Health Research Health Technology Assessment Prioritisation Committee: Integrated Community Health and Social Care Panel (A) and a member of Methods Group (A).

Grant information: This study was funded by the National Institute for Health Research (NIHR) Research for Patient Benefit Programme [PB-PG-1217-20012]. Professor Salisbury is a National Institute for Health Research (NIHR) Senior Investigator. The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care. *The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.*

Copyright: © 2022 Murphy M *et al.* This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

How to cite this article: Murphy M, Salisbury C, Scott A *et al.* **The person-based development and realist evaluation of a pre-consultation form for GP consultations [version 2; peer review: 2 approved]** NIHR Open Research 2022, **2**:19 https://doi.org/10.3310/nihropenres.13249.2

First published: 17 Feb 2022, 2:19 https://doi.org/10.3310/nihropenres.13249.1

REVISED Amendments from Version 1

Further expansion on sampling strategy, expansion on CMOCs 1 and 3, and further discussion on the similarities and differences with online triage forms. Other minor changes including correction of typographical errors were made throughout.

Any further responses from the reviewers can be found at the end of the article

Plain english summary

The problem

For some patients, GP consultations are too short. Sometimes patients' problems are missed. We wanted to improve GP consultations.

What we did

We developed a better way to start and end consultations using a new digital method. Before a GP consultation, patients fill in a form online that lets them describe their problems. This is shared with their GP. At the end of the consultation GPs can give patients a one-page summary of what was discussed. This paper reports on how we developed and tested the online pre-consultation form.

How we tested it

We first piloted the online form in three GP practices in turn. We interviewed patients, GPs and an administrator in each practice and made changes based on their suggestions. Each new version was tested in a new practice.

We then tested the final online form together with the one-page summary in four practices with 30 patients each. We interviewed patients and GPs to find who it was most useful for and when.

What we found

GPs and patients agreed the final version of the online form was much better than the first version. The percentage of patients who filled in a form after getting a text message increased from 15% in practice 1 to 50% in practice 3.

By testing the final form in four practices, we found it worked best for people who find technology easy to use and have hidden concerns or worries about the consultation. These patients found their GP was better prepared and problems were dealt with better. GPs thought consultations were more efficient.

Conclusions

The approach we took was very successful in developing the online form and patients and GPs found it useful. The technology would need to be improved before it could be rolled out more widely.

1 Background

1.1 Rationale for study

Patients often leave GP consultations with unaddressed concerns^{1,2}. This can lead to high rates of re-consultation and increased morbidity in the population. Previous research shows that approximately 27% of patients consulting in primary care

have seen a doctor or nurse for the same problem in the last four weeks³, and more recently published research demonstrates that up to 50% of consultations in primary care are followed by another consultation within two weeks⁴. Although there are no estimates of re-consultation for unaddressed concerns in primary care, we know that problems are missed in up to 50% of primary care consultations², and that reducing consultation rates by just 1% in 2016 could have saved the NHS over £100 million⁵. Primary care patients often present with multiple complex problems, many of which are unrelated to physical symptoms, and include informational needs on symptom-management or self-care, emotional problems, health concerns or social problems⁶. In the context of multiple presenting problems, GPs tend to focus on physical symptoms7. While this prioritisation is entirely appropriate to ensure correct diagnosis and patient safety, any missed opportunities to improve patient understanding and ability to self-care are also costly: a study in 2015 found that increasing patient engagement in their own health could save the NHS £2 billion by 20208. Small changes to improve the ability of GPs to thoroughly and efficiently address patients' presenting problems, concerns and questions could therefore have considerable impact on the overall NHS budget as well as on patient and doctor well-being and satisfaction.

The Calgary-Cambridge guide, which is used as a basis for training medical students and doctors, identifies six steps to conducting a GP consultation: initiating, information gathering, providing structure, relationship building, explanation/planning and closing². Opportunities to address patients problems are commonly missed at consultation initiation (when the GP should elicit the patients' reason for attendance)². Problems can remain unaddressed at consultation closure, if advice given is unclear, particularly with regards "safety-netting": i.e. advising patients what to do if the problem does not resolve, or gets worse9. Research suggests that interventions at each end of the consultation can help to address patient concerns. At consultation initiation, sharing the results from patient-reported outcome measures (PROMs) with clinicians can help to elicit concerns¹⁰. At consultation closure, providing the patient with written information as well as spoken can improve recall and adherence¹¹.

To help with this problem, we designed an intervention: the Consultation Open and Close (COAC) intervention which used a patient-filled pre-consultation form, completed prior to and discussed at consultation opening, and a summary report provided by the doctor at consultation closure. We then tested these interventions in a feasibility study.

1.2 Review of Evidence of pre-consultation forms *1.2.1 Consultation initiation: eliciting all concerns*

Active listening was described by Carl Rogers as absorbing everything a person says without "subtracting" or "amending"¹². Many patients regard the ability to listen as the single most important characteristic of a good doctor¹³. The importance of active listening has long been recognised and incorporated into undergraduate medical curriculae¹⁴.

Despite this, studies have shown that GPs often interrupt patients, particularly during the patient's opening statement (or patient

monologue)². Although GPs may perceive that the patient monologue is wasting time, in fact, it takes only 30 seconds on average¹⁵. One study showed that doctors waited an average of 23 seconds before interrupting the patient's opening statement¹, when less than ten seconds more would usually allow the patient to finish. When GPs interrupt, they are nearly always doing so with their patients' interests in mind: recognising the importance of listening, but having limited time to gather essential information from patients before moving onto diagnosis and advice16. In many cases, GPs interrupt because a patient is providing medical history which the clinician already knows. One approach to dealing with this problem is "physician goes first": whereby the doctor starts the consultation with a very short synopsis of what he/she knows about the patient's recent medical history, before asking the patient about their goals for consulting and allowing them to speak uninterrupted¹⁷.

This approach can be facilitated by a review of the patient's medical record before the start of the consultation, and also by patient completion of a PROM, which is shared with the clinician before the consultation. This can save valuable consultation time, by giving the GP or nurse an immediate oversight of the patient's current state of health and immediate presenting problems¹⁸.

1.2.2 Use of electronic PROMs in primary care at an individual level

PROMs were originally designed for use at aggregate level, to compare the scores of groups of patients receiving different care¹⁰. However, PROMs are increasingly being used at an individual-level to inform a consultation, set priorities or aid diagnosis10. Feedback of individual-level PROMs information to clinicians has been used most widely in oncology¹⁹. It has a positive effect on patient experience and patient care by promoting patient self-reflection thereby helping patients remember their main concerns²⁰, by improving patient-clinician communication²¹ and by making it easier for patients to share information which they find it difficult to express verbally²². There is less evidence for an impact on outcomes¹⁹. Trials of PROMs feedback to clinicians which have shown effects on patient outcome tend to use randomisation at the physician or practice level, rather than the patient-level, i.e. with each practice randomly assigned to using PROMs feedback rather than each patient²³.

A realist review of feedback of individual-level PROMs to clinicians found that one mechanism by which they can work is by raising clinicians' awareness of patient concerns¹⁰. In the context of increasing GP workload, it is important that these PROMs capture relevant information, delivered succinctly. Benefits of electronic PROMs (ePROMs) include; remote completion, instant transfer, and filtering and summarising of data so clinicians see only the most important information. They also solve problems with questionnaire completion in waiting rooms; most primary care patients book an appointment only one or two days in advance so recruiting patients before a consultation normally requires waiting room recruitment²⁴. This limits the time for form completion, and some patients will be called in to their consultation before completing the questionnaire²⁵.

The current widespread digitisation in general practice²⁶ offers a timely opportunity to integrate an ePROM into clinical practice for use at an individual-level to help identify patient concerns.

Electronic triage forms were mandated by the NHS long term plan²⁷ and were rolled out across general practice during this study. Electronic triage forms have features that are common to ePROMs completed before consultations; they both collect clinical information from the patient which is shared asynchronously with a clinician. However, they differ in purpose and content. Electronic triage forms are primarily used as a triage tool and collect information on symptoms. The patient may not receive a consultation after completion of an electronic triage form, but can be advised to self-care, go to a pharmacist or Emergency Department (ED) or receive advice from the GP through email or the triage portal. The primary purpose of ePROMs shared with clinicians before consultations is not triage, as the patient has already has a booked appointment; it can serve multiple purposes, include uncovering additional information or providing more detail on a patient's problems¹⁰.

The aim of this study was to develop and test an intervention to more comprehensively address patients' concerns in general practice through use of a pre-consultation form discussed at consultation opening and provision of a summary report on consultation closure.

This paper describes the development of the pre-consultation form and a realist evaluation of its use from the feasibility study of the COAC intervention. The development and testing of the consultation summary report and the full feasibility study results are reported separately in two papers published alongside this one.

2 Methods

2.1 Study setting

This study was based in primary care involving general practices serving different patient populations in Bristol, North Somerset and South Gloucestershire (BNSSG). Practices were selected from areas within a range of socioeconomic deprivation levels as well as urban, suburban and rural areas.

The COVID-19 pandemic occurred six months into this study. Under an NIHR directive, the study was paused in March 2020 and restarted in September 2020. Research protocols were updated so the intervention and research did not require face-to-face contact.

2.2 The Consultation Open and Close Study

The COAC Study involved the development and testing of an intervention, incorporating use of an individual-level PROM at consultation opening and written information at consultation closure. The primary aim of the COAC study was to develop and test the feasibility of a complex intervention designed to more comprehensively address patients' concerns in general practice, thereby reducing re-consultation rates, improving patients' well-being and health knowledge, reducing health concerns and increasing patients' confidence in their health provision and health plan.

The COAC study incorporated two phases: an Intervention Development study (Study 1) and a feasibility study (Study 2) as follows:

Intervention development study: This involved design of a complex intervention to improve the ability of GPs or nurse practitioners to address patients' concerns through a) development and testing an electronic patient questionnaire at consultation opening and b) developing and testing a summary report at consultation closure, which is either printed or texted to the patient or is accessible from the patient record. These were designed and evaluated separately, in accordance with MRC guidance for design of complex interventions²⁸.

Feasibility study: In this study, the COAC intervention was tested in a cluster-randomised framework to establish the feasibility of both a randomised-control trial of the intervention and the intervention itself.

The sequential nature of the studies is shown in Figure 1.

This paper describes the development of the pre-consultation form and a realist evaluation of its use from the feasibility study of the COAC intervention. The development and testing of the consultation summary report and the full feasibility study results are reported separately in two papers published alongside this one.

2.3 Recruitment

2.3.1 Practice recruitment

Practices were approached by the NIHR Clinical Research Network for the West of England (hereafter referred to as the CRN) with the information on the study. Practices were recruited to the two phases separately; with practices who participated in the Intervention Development study actively encouraged to continue their participation in the Feasibility study.

For each study, the CRN shared the study Research Information Sheet for Practices (RISP) with practices who met the inclusion criteria. Interested practices then contacted the study chief investigator (CI) who arranged a meeting(s) with the practice manager, or GP research lead.

Practice representatives were asked to sign a practice agreement consenting to the practice taking part in the study. Practices were approached for Study 1 in November 2019 (three practices were required for Study 1); and for Study 2 in May 2021 (six practices were required for Study 2).

All selected practices already used SMS software (MJOG and accuRx) and the patient record system EMIS. Administrators were expected to be familiar with the process of sending batch texts using practice SMS software (e.g. MJOG) and in uploading reports to and setting alerts in EMIS.

2.3.2 Eligibility criteria

For the Intervention Development study (three practices) we purposively selected: one practice in the top deprivation quartile, one at the median, and one in the lower quartile. For the feasibility study (six practices), we selected three practices in the top two deprivation quartiles and three practices in the bottom two.

Patients in both studies were included who were:

- Aged 17 or over (on date of SMS invitation to participate)
- Had an upcoming appointment with a recruiting GP within the next week.



Figure 1. The Intervention Development and Feasibility Studies.

Patients were excluded if they were:

- Housebound
- Had not given permission to receive SMS messages from the practice
- Had a recent diagnosis of life-limiting or life-threatening illness
- Were deemed by the GP to be at serious suicidal risk
- Were unable to complete questionnaires in English even with the help of carers.

2.3.3 Sampling criteria

Participants were selected for interview on the basis that they had completed a pre-consultation form. Within this, patients were purposefully sampled based on the information on their forms; we sought to include some patients who had single and some with multiple problems; some patients who provided a lot of detail and some who provided no detail; and some with red/amber flags and some without. Because the recruitment process was designed to be as straightforward as possible for patients, we did not collect any demographic information and were therefore unable to use characteristics such as gender, age or ethnicity to purposefully sample patients.

The remainder of the methods are described separately for the Intervention Development study and feasibility study in the respective two sections that follow.

2.4. Intervention development study methods *2.4.1 Approach*

The Intervention Development study was carried out in two distinct parts, one for development of the online pre-consultation form and one for development of the summary report provided at consultation closure. Development of the pre-consultation form is described in this section. A prototype was developed based on the research literature and a series of patient and public involvement (PPI) group consultations. This was then tested with actual patients using a person-based approach, which involves using mixed-methods research to systematically investigate the needs, attitudes and situation of the people who will be using the intervention²⁹. Through the person-based approach, each step of the intervention was tested in rounds and adjusted after each round according to the feedback given from patients and clinicians. This iterative approach is shown in Figure 2.

2.4.2 Prototype development Starting position

Pilot work with a PPI group suggested that the pre-consultation form should include both individualised information (a list, generated by the patient, of their reasons for attending, and the key issues they would like to discuss) and standardised information (a short list of questions on common problems, with tick-box answers). A standard questionnaire and report were developed before the study commenced, based on the Primary Care Outcomes Questionnaire (PCOQ) and this was used as the starting point for person-based development and testing.

Activity:	Pre-consultation form person-based testing (Nov 2019 – Oct 2020)
РРІ	A PPI group were consulted on the initial pre-consultation form and feedback
consultation	report.
Recruitment	Three practices were recruited.
•	
Set-up and Training	Admin staff were trained to send SMS messages to patients with a link to the pre-consultation form, and to upload the summary report from REDCap ³⁰ to the patient record system. Two GPs in each practice were trained in using the report.
Person-based testing	GPs tested the system with 15 patients per practice.
Iterative evaluation and refining	The GPs, admin staff and a targeted six to seven patients per practice were interviewed at the end of each phase. The process, and respective technology was adjusted after each round, in accordance with the iterative nature of the person based approach. The process was then tested in the part exercise.

Figure 2. Person-based approach taken to develop the pre-consultation form.

The PCOQ is a validated generic questionnaire which was developed to capture the main outcomes which can be influenced by primary care. It has 24 items which include physical and emotional symptoms and function, self-care, health behaviour, adherence, and a sense of support^{25,30}.

The pre-consultation form was put into an online survey using the University of Bristol database system REDCap: a low-cost, secure, web-based electronic data capture system for clinical research³¹. Only 18 of the 24 PCOQ items were included, as six items refer to the patient's confidence in seeking healthcare, and are not suitable for sharing with their clinician. Versions were developed for smartphone and computer.

A process was designed for the information from the pre-consultation form to be downloaded from REDCap and attached to EMIS in a pdf report format for the GP or nurse to review before the consultation. Rather than simply attaching the full questionnaire, this was formatted so that it was short and easy for clinicians to digest. It contained two sections: an individualised section with the patients' reasons for attending, and a standardised section, which was a colour-coded list of responses to standard questions.

Initial programme theory

An initial programme theory of how the COAC intervention was intended to produce outcomes was designed. This was drafted by the study CI and reviewed by the study co-investigators and PPI group before finalisation. This is shown in Figure 3.

2.4.3 Data collection / measures

Data collected in the Intervention Development study included clinician and administrator questionnaire data and qualitative interviews. Interviews were carried out by MM and AS and audio-recorded. We aimed for 20 patient interviews, six GP interviews and three administrator interviews.

The purpose of these interviews was to inform development of the intervention through a person-based approach (which takes place in rounds, with the intervention amended at the end of



Figure 3. Proposed initial programme theory of COAC.

each round). Topic guides therefore focused on the feasibility and perceived usefulness of the pre-consultation form and on the proposed design of the intervention. Patient and GP topic guides are available as open access data (see data availability statement).

Prior to the COVID-19 pandemic, some interviews were conducted face-to-face in the patients' own homes or other location of their choice, and GP/administrator interviews in the relevant health centre. After March 2020 interviews were conducted by telephone or video link.

2.4.4 Analysis

.

Interviews were transcribed and analysed at the end of each round. The analysis focussed on establishing what changes were required to the pre-consultation form in that round before testing again in the next practice.

To do this, "guiding principles" were established. These are fundamental to the person-based approach, and highlight the objectives of the intervention and the key features that will address context-specific behavioural issues in support of these objectives²⁹. The guiding principles were drafted by the CI, adjusted by the PPI group and agreed by the study co-investigators. A coding framework for changes identified was then established (see Table 1). This framework contained codes to identify the reason for making each change, with reference to the guiding principles and the initial programme theory. After each interview was transcribed, one of the researchers (the qualitative researcher or the CI) listed the possible changes arising from that interview and categorised these possible changes according to the list shown in Table 1. The second

researcher then checked this and, if necessary, added new changes to the list, or modified existing ones. The two researchers then discussed any areas of disagreement.

At the end of each round, the co-investigators all reviewed the table of changes and a final list for the round was agreed. The changes were implemented and the revised pre-consultation form was taken forward to the next round.

This continued for three rounds until a final version of the pre-consultation form was agreed.

2.5 Patient and public involvement

This research was informed by PPI both before the study commenced and during the study. PPI contributors received expenses and reimbursement in line with INVOLVE guidance³².

Pre-study PPI was carried out with an existing group that had formed for another study on improving care for patients with long-term conditions³³. The pre-study PPI group met three times before the study commenced to advised on the proposal, the process, the pre-consultation form and the consultation summary report. This group was instrumental in the decision to use the PCOQ as the basis for the pre-consultation form and include an option for patients to provide their reasons for consulting.

A new PPI group was convened at the start of the study to include a more diverse membership, involving different ethnic groups and people both with and without long-term conditions. The group then met five times throughout the study. Two of these meetings were specifically to design the pre-consultation form as follows:

Table 1.	Coding	framework for	lable of	Changes.	

.

6.01

Coding	Coding framework							
Code	Stands for	Means						
IMP	Important for intervention uptake and effectiveness	This is an important change that is likely to impact intervention uptake or effectiveness or is a precursor to that (e.g. acceptability, feasibility, persuasiveness, motivation, engagement), and/or is in line with the programme theory and/or is in line with the Guiding Principles.						
EAS	Easy and uncontroversial	An easy and feasible change not involving any major design changes. For example, a participant was unsure of a technical term, so a definition is added.						
REP	Repeatedly	This was said repeatedly, by more than one participant.						
EXP	Experience	 This is supported by experience, for example: PPIs agree this would be an appropriate change. Experts (e.g. clinicians on your development team) agree that this would be an appropriate change. Literature: This is supported by evidence in the literature. 						
NCON	Does not contradict	This does not contradict experience (e.g. evidence), or the programme theory, or the Guiding Principles						
RES	Research relevant	This is a change to the design of the research, not the intervention						
NC	Not changed	It was decided not to make this change. Please explain why (e.g. it would not be feasible; or only one person said this).						

Meeting 1: In the first meeting, members were introduced to the study and made suggestions on the overall design. They suggested recruiting a range of GP practices with different approaches to appointment booking and extending the study to include Nurse Practitioners.

Meeting 2: In the second meeting, members gave detailed input to the pre-consultation form and report before the person-based development started. This resulted in substantial changes to the pre-consultation form which are captured in the Table of Changes (see extended data) and in the summary of changes (section 3.1.2).

Some members of the PPI group raised concerns that the study could increase health inequity. They were concerned that the recruitment process for the intervention depended on patients firstly having access to a smartphone or computer, and secondly having the ability to complete a questionnaire on smartphone or computer. Despite these concerns, most PPI members felt that proceeding with a digitally-based intervention was acceptable, since if the intervention was useful, work could then be done on improving and extending access.

2.6 Feasibility study methods

To provide some context for the realist evaluation, brief details are provided in this paper on the randomisation, recruitment and consent and data collection / measures of the feasibility study. This section mainly focusses on describing the data and analysis used for the realist evaluation embedded within the feasibility study. The full feasibility study results, including recruitment rates and suggestions for improving these is discussed in the feasibility study linked paper³⁴.

2.6.1 Randomisation recruitment and consent

In the feasibility study, both the pre-consultation form and summary report were used together in six practices, four randomised to intervention and two to control. Practices who had participated in the Intervention Development study were approached by the CI and new practices were approached by the CRN. Each practice was asked to recruit 30 patients, resulting in 120 in the intervention and 60 in the control (see Table 2).

General practice administrators searched their practice database using an electronic search strategy which identified patients with upcoming appointments who met the inclusion criteria and sent batch SMSs to patients with a link to a baseline questionnaire hosted on REDCap. The baseline questionnaire consisted of both the pre-consultation form for sharing with the GP and additional questionnaire data required to establish the patient's pre-consultation state of health for evaluating the intervention. The SMSs contained the patient EMIS number

Table 2. Patient recruitment target incontrol and intervention practices.

	Intervention	Control	Total
Practices	4	2	6
Patients	120	60	180

and the patient needed to input this so their questionnaire could be identified.

Administrators received an alert when a patient completed a questionnaire. On a regular basis, the administrator downloaded the summary report from REDCap to pdf and attached it to the EMIS patient record system. The baseline questionnaire included an information screen explaining the purpose of the study and how the data would be used. Return of the questionnaire indicated consent to participate in the study. Patients were explicitly asked to consent to their contact phone number being shared with the University of Bristol for the purposes of sending a follow-up questionnaire. Further consent for use of that phone number to contact the patient for interview and for access to the patient's record for demographics and re-consultation rates was requested in the follow-up questionnaire³⁵. A similar approach has been taken for a number of other cluster trials³⁵⁻³⁷. The researcher then took informed consent for recording the interviews and use of anonymised quotations in publications prior to the interview itself. This consent was written for face-to-face interviews and audio-recorded for telephone interviews. Before the start of the COVID-19 pandemic, all consent was written. The ethics committee approved an amendment to collect audio-recorded consent for patients interviewed during the pandemic. A workflow for this is shown in Figure 4.

2.6.2 Data collection / measures

Feasibility study data included clinician and administrator questionnaire data, interview data, and quantitative patient data. The quantitative patient data is described in the feasibility study linked paper³⁴. Interview and questionnaire data was collected as follows:

Clinician questionnaire data: The GP questionnaire requested information for each consultation (new/review), modality (face-to-face, telephone or video), whether the pre-consultation form was useful, and why a summary report was used.

Interview data: Interviews were conducted by the CI and the project research associate. Topic guides were designed to inform a realist evaluation and therefore focused on the outcomes that patients/GPs perceived, the mechanisms by which these were achieved and the contexts. Patients and practitioners were interviewed to the point of achieving "theoretical sufficiency", i.e. when the data analysis has yielded one or more coherent theories which are relevant to the study aims³⁸. Interviews were conducted by phone and audio-recorded.

2.6.3 Analysis

Realist evaluation seeks to explain the complex relationship between context, mechanisms and outcome. The explanatory proposition of realist evaluation is that interventions work (i.e. have successful outcomes) only in so far as the individuals involved take up ideas and opportunities (mechanisms) within the social and practical conditions in which they are operating (contexts)³⁹. This is then reported in terms of contextual factors (What elements of the intervention work, for whom, in what consultations?) and content-mechanism-outcome configurations (CMOCs). A CMOC is a hypothesis that the program works



1. SMS messages were to patients with upcoming booked appointments with recruiting GPs.

2. Reports were attached to EMIS and an EMIS flag set for records with attached report

3. A sub-set of patients (based on pre-agreed criteria, but ultimately selected by the GP) were given a summary report

4. A follow-up text was sent from REDCap to patients in both arms

Figure 4. feasibility study workflow: intervention and control arms.

to produce an outcome (O) because of the action of some underlying mechanism (M), which only comes into operation in particular contexts. (C) $\,$

The realist evaluation used the interview data collected in the feasibility study supplemented with the interviews from the intervention development study. To carry out the realist evaluation, the CI (MM) read and re-read the initial interview transcripts from both patients and practitioners, in order to gain an overall view of the accounts given and to identify patterns in the data. She then revised the programme theory and devised an initial set of CMOCs. The research associate (AS) independently developed three lists of context, mechanisms and outcomes. These were cross-checked against the CI's CMOCs which were then revised and detailed evidence presented against each of them. An experienced realist evaluator (GW), then read through the detailed evidence and the final CMOCs were agreed in collaboration. Four researchers (MM/AS/GW/CS) reviewed the realist evaluation and programme theory before finalising.

2.7 Sponsorship, funding and ethical arrangements

This study was sponsored by the University of Bristol. Ethics approval was granted by Frenchay Research Ethics committee⁴⁰ and the Heath Research Authority (HRA). BNSSG Clinical Commissioning Group Research and Evidence Team provided research and development approval. The study was NIHR

funded and supported by the NIHR Clinical Research Network who liaised with centres on the researchers' behalf.

Insurance was provided by the University of Bristol as research sponsor. The study sponsor and funders did not have any role in study design; data collection, management, analysis, and interpretation of data; writing of the report; or the decision to submit the report for publication.

The Feasibility study was registered in the ISCTRN registry (ISRCTN13471877) and on the CRN portfolio (42005). The study protocol was published before recruitment completed⁴¹.

3 Results

3.1 Intervention development

3.1.1 Participants

Table 3 shows the number of recruits to the Intervention Development study. Three practices were recruited from the top, middle and bottom of the index of deprivation (IMD) score, where 1 indicated a high level of deprivation and 10 a high level of affluence. Each practice had two participating GPs. One of these was an advanced nurse practitioner, but for simplicity has been referred to throughout as a GP. Each practice recruited their target of 15 patients, which was 6 to 9 per GP. We had intended to interview 20 patients from the 45 recruited but were only able to interview 12.

Recruiting practice	Practice IMD score	Date	# Recruiting GPs	# Admin staff involved	Patients recruits per practice (target = 15)	Patient interviews per practice (target = 6 to 7)	Response rates (SMS sent / Patient recruits)
Practice 1	9	Dec 19	2	2	15	6	15%
Practice 2	5	Feb 20	2	1	15	2	17%
Practice 3	1	Nov 20	2	2	15	4	50%
Total			6	5	45	12	

Table 3. Development of the pre-consultation form: GP and patient recruits.

3.1.2 Summary of changes

The person-based approach relies on a set of guiding principles. These were agreed in advance and informed the intervention development by highlighting the objectives of the intervention and the key features that will address context-specific behavioural issues in support of these objectives²⁹ (Table 4).

Table 5 summarises the key elements of the pre-consultation form that were changed over the 3 rounds. As shown in Table 5, the form was shortened during the process and the wording clarified. Minor wording changes were added to improve uptake and encourage patients to add more free text. Diffusion of innovation theory was used to inform this. The report instructions and training for GPs were adjusted to emphasise how important it was for GPs to let the patient know they had read and understood the report. Administrative processes were simplified.

Table 5 is a summary of 32 changes documented, agreed and made during the three rounds. The full table of 32 changes, including verbatim quotes and coded rationale for making each change are available as open access data (see data availability statement).

Figure 5 and Figure 6 show the initial (start of round 1) and final (end of round 3) pre-consultation report seen by GPs.

Figure 7 to Figure 12 show the final pre-consultation form from the patient point of view. The patient receives an SMS on their phone with a link to the form and their unique ID number to enter in the first screen (Figure 7). This form lets the patient know that completion is optional but should improve their appointment experience. On clicking the link, the patient has to enter this ID number before proceeding (Figure 8). The patient then sees a screen of information about the study. Again, this screen encourages the patient to complete the form by saying other patients have found it useful. It also contains a link to the study website for more detailed information (Figure 9). If the patient clicks Next, they are asked if they are completing the form by phone or computer (Figure 10). Depending on what they select, they are shown a matrix version of the form (computer) or one item per screen (phone). The first screen asks for the reasons for consulting and how long the problems have persisted (Figure 11). There is then one screen for each of the 13 remaining questions. Each question has the same 5-point categorical response scale. Figure 12 shows the phone version of the first question for "pain". If the patient

responds with anything other than "not at all", a free text box appears asking them for more information. This question populates the first row in the report that goes to the GP (shown in Figure 6). The remaining 12 questions follow a similar format and are also used to populate the report, with the extremity of the response given by the patient driving the colour of the row.

The final agreed administrative process is shown in Figure 13.

- **1.** The administrator runs a pre-built report in EMIS which identifies patients with upcoming appointments with recruiting GPs.
- 2. The administrator exports the list of patients to a csv file.
- **3.** The administrator opens MJOG, imports the list of patients and sends an SMS with the link to each patient.
- **4.** The patients receive an SMS with a link to the pre-consultation form on their phone.
- 5. **Recruited patients** give consent, complete the form and the data is stored in REDCap.
- 6. The administrator exports data from REDCap to a csv file.
- 7. The administrator runs a macro to generate an individual pdf file for each patient.
- 8. The administrator attaches each pdf file to the patient record and adds a note to the appointment book for that patient to alert the GP that the report is there.
- **9.** The GP reviews each patient report on EMIS before the consultation and carries out the COAC intervention with that patient (summarising the information for the patient, asking if there is anything else, listening without interruption, providing a report on closure for a subset of patients).

3.2 Realist evaluation

3.2.1 Participants

Forty-five interviews were carried out in the feasibility study: 30 patients, nine GPs and six administrators. In addition, the eighteen GP and patient interviews from the intervention

Table 4. Guiding principles – pre-consultation form.

Intervention Design Objectives	Key Features
To make the pre-consultation form appealing for patients to complete	Convincing text message and initial screenTargeted at the right kind of patientUsing positive language throughout
To create a positive and beneficial experience for patients completing the pre-consultation form	 Easy to complete Seems relevant to patient Ensuring the intervention provides something interesting, relevant, and helpful for the user (patients)
To make the report as useful as possible for GPs	 Relevant information on the form Clear and easy to read form Access and reading of the form fits within normal process of the consultation Patients provide optimal amount of info and the right kind of info
To create a positive and beneficial experience in the consultation, where the pre-consultation form promotes communication and problem solving	 GP training and ongoing support Ensuring the intervention provides something interesting, relevant, and helpful for the user (patients and GPs) Reciprocating intervention usage by providing immediately rewarding feedback

Table 5. Summary of changes: pre-consultation form.

	Issue identified	Feature added		
PPI / patients	The validated questionnaire used (PCOQ ⁴²) was too long and repetitive. It felt like a research tool, not for practical use.	The questionnaire was simplified and reduced in length. Some of the question wording was adjusted to gear it more towards practical use.		
PPI / patients	Further information on each question is asked for at the end of the form. It would flow better if this was asked for throughout.	The form was adjusted so that additional questions asking for more information appeared immediately a patient gave a low score.		
Patients	It was very important to patients that GPs mentioned they had read the report.	Emphasised this in training, and added "remember to tell the patient you have read this report"		
Empirical	In the first practices, just over 15% of patients who were sent a message responded.	After the first two practices, added into the SMS: "Completing this form should really improve your experience of your appointment." (diffusion of innovation theory technique)		
GPs	The report is more useful when the patient adds free text, especially for health concerns and support.	Final version flagged to patients those questions where it was really important the patient put in additional information.		
GPs	To simplify the report for GPs, patient responses were summarised and some were merged. Some meaning was lost in this: e.g. "adherence to medication" and "adherence to healthy lifestyle" are two distinct ideas which it was confusing to merge on one line.	The final report contained 10 lines. Most pertained to a single question, and the captions for these, although summarised, attempted to follow the language of the question patients responded to. Three pairs of questions were still merged, but GPs confirmed these were clear.		
GPs	It was not fully clear when the free text is system generated and when it is written by the patient	Added a line to GP report "Blue signifies free-text written by the patient. Black text is based on patient's selected responses."		
GPs	Patients sometimes give multiple reasons for attending all on one line, which is difficult to read. Some are existing problems and some new, so it would be useful for patient to indicate if and how many times they have consulted about a problem before.	Ask patients for their reasons for attending at the start of the form, stating that they will be able to enter up to three reasons (so that they are split out into separate lines for each reading). Also allow patients to indicate whether they have consulted about the problem before		

	Issue identified	Feature added
GPs	GP raised in training sessions that some of the reports are very detailed and might be too time-consuming to read.	In practice, reading a report took about 20 secs. A timed reading to demonstrate this was incorporated into the GP training.
GPs	Initially, administrators sent a task to GPs and added a pop-up. GPs felt this was unnecessary and would prefer a note in the appointments book.	Agreed that the words "COAC patient" should be added as a note in the appointments book after each COAC Study patient appointment
Admin	The administrative process of generating an individual link for each patient was time-consuming	The process was adjusted so that patients input the EMIS number themselves. Administrator then sent the same link to each patient.
Admin	Some patients completed the form immediately before their appointment, and admin did not have time to add it to the record.	Added into the SMS: Please complete the form today so that your GP/nurse has time to read it. Sent a notification to the administrator each time a report is completed on REDCap

Your patient gave the following reasons for attending:

- 1 Indigestion and heatburn
- 2 Night cough
- 3 Really sore left foot when walking

The list below highlights some other problems your patient might have.

Green indicates no problems. Light green indicates slight problems. Amber is moderate and red is severe.

1 Pain	(1)	3	e	Moderate	The pain is in the foot and only there when walking. I've had it for a month.
2 Other Physical Symptoms	(1)	2	:	Slight	Cough and heartburn. Have had both for about two months. They mostly come at night.
3 Effect on Daily Life	(1)	2	٢	Slight	I'm not sleeping with the cough and pain and I can't walk far and have stopped playing tennis.
4 Low mood / Anxiety	(2)	2	٢	Slight	Not low in mood or depressed. Slightly Anxious or stressed.
5 Health Concerns	(2)	5	8	Extreme	Moderately worried symptoms might indicate a serious illness.
	-				
6 Health Knowledge	(2)	3	e	Somewhat lacking	Somewhat lacking in understanding of current health problems. Not lacking knowledge about how to prevent future health problems.
7 Support	(2)	2	٢	Slightly lacking	Face to face conversation would help
8 Confidence in health plan	(1)	1		Extremely confident	
9 Adherence to advice	(2)	3	e	Some advice	l don't think any of the medicine l've taken is working for the cough. I want to find out what it is.

Notes to GP:

At the start of the consultation:

1. Recap the reasons the patient put that they were here - e.g. "I see from your pre-consultation form you want to ask about the flu injection."

Mention other severe or moderate problems raised in the questionnaire e.g. "I also see you are in quite a bit of pain and are worried about it"
 Ask for more information "what is it you want to discuss today"

Figure 5. Pre-consultation report: Pilot version (start of intervention development).

Your patient gave the following reasons for attending:

1	Indigestion and heatburn	consulted about 1 - 2 times
2	Night cough	new problem
3	Really sore left foot when walking	new problem

Th Gro	ïhe list below highlights some other problems your patient might have. Green indicates no problems. Light green indicates slight problems. Amber is moderate and red is severe.					
1	Pain	(1)	3	e	Moderate	The pain is in the foot and only there when walking. I've had it for a month.
2	Other Physical Symptoms	(1)	2		Slight	Cough and heartburn. Have had both for about two months. They mostly come at night.
3	Effect on normal activities	(1)	2	٢	Slight	I'm not sleeping with the cough and pain and I can't walk far and have stopped playing tennis.
4	Low mood / Anxiety	(2)	2		Slight	Not low in mood or depressed. Slightly Anxious or stressed.
5	Concerns about serious illess	(1)	5	8	Extreme	Moderately worried symptoms might indicate a serious illness.
6	Health Knowledge	(2)	3	()	Somewhat	Somewhat lacking in understanding of current health problemss.
	neuth thowledge	(2)			lacking	Not lacking knowledge about how to prevent future health problems.
7	Support	(2)	3	•	Somewhat lacking	Face to face conversation would help
8	Adherence to treatment	(1)	1	٢	All advice	
9	Healthy lifestyle	(1)	1	٢	All of the time	
10	Confidence in health plan	(1)	3	•	Moderately confident	I don't think any of the medicine I've taken is working for the cough. I want to find out what it is.

REMEMBER TO TELL THE PATIENT THAT YOU HAVE READ THIS REPORT. PLEASE ALSO MENTION THE FOLLOW-UP. *Blue signifies free-text written by the patient.* Black text is based on patient's selected responses.

Figure 6. Pre-consultation report: Final version (end of intervention development).

Dear Mr Clark, Please click on the link to open a form to fill in before your upcoming GP appointment, first noting your unique ID number 1100464

https://is.gd/CpATRf

Completing the form is optional but should improve your appointment experience. Please complete the form today so your GP has time to review it.

GP Practice Name

shown ir	er the unique ID humber that wa rour text message
* must pro	de value
	Continue

Figure 8. Pre-consultation form: Opening Screen.

We sent you this message because you have a doctor or nurse's appointment soon. Doctor's appointments are only 10 minutes long. This sometimes makes it hard to explain everything you need to. Your practice is working with the University of Bristol to try out a new online form that tells your doctor or nurse about your health and the reason for your visit before you go. This is a tried and tested form which other patients have found really improves communication with the GP.

It's up to you if you want to fill in the form. All your personal details will be kept private and can only be seen by staff at your practice. Information from the forms will be stored on secure University of Bristol servers and will be used by researchers at the University but they won't be able to see your name or any of your personal details. If you want more information about how your data will be stored you can find it on the study webpage http://www.bristol.ac.uk/coacstudy/

The online form has 14 questions and should take about 5 minutes to fill in. If you would like to participate, please click on the Next button.



Figure 9. Pre-consultation form: Information and consent screen.

Information to inform your consultation	AutoScroll On Resize font: ⊕ ➡
Please click on the phone butt using a phone, or click on the o button if you are using a comp	on if you are computer outer.
When you click on phone or co will be taken to the online forr can start filling it in.	mputer you n and you
Phone	
Computer	
	reset

Figure 10. Pre-consultation form: Selection of phone or computer.

question number 1 of 26
Please enter you ₁₄ hain reason or reasons for making an appointment with the doctor (you will be able to enter up to three reasons)
Reason 1
Sore throat
Have you consulted a GP or nurse about this before?
Yes: once or twice 🗢
Reason 2
Reason 3
Next
16561

Figure 11. Pre-consultation form: Reasons for consultation screen.

uest	ion number 2 of 214	
At Ho Pa	t he moment: w much are you <u>currently affected</u> l in or discomfort	ру:
	😲 Not at all)
	Slightly	
	Moderately	
	Quite a bit)
	Extremely	
		reset
PA Yo ab ha ple ne	IN u just indicated that you are bother in. If you would like to say anything out this (where the pain is, how long s lasted, if it is constant or occasion ease put it below. Otherwise go to th xt question.	ed by else g it al) ne
)
	Next	J
		reset

Figure 12. Pre-consultation form: Pain question.



Figure 13. pre-consultation report generation: process diagram.

development phase were also used to inform the realist analysis. Interviews at each site in each phase are shown in Table 6.

In the qualitative analysis which follows Patients 1 to 20 are from the intervention development study and patients 30 to 50 from the Feasibility study. So that the evolution of their views can be compared, the same identifier is used across the studies for GPs who were in both studies.

3.2.2 Summary findings from the process evaluation

The process evaluation of the feasibility study is presented in the linked paper³⁴. Although this is a separate publication, some key findings are briefly summarised here to provide context. A key finding of the feasibility study was that the pre-consultation form and summary report are useful for different types of patients and consultation and each intervention results in different outcomes, triggered via separate mechanisms. It was therefore more appropriate to carry out a separate realist evaluation for the pre-consultation form and the summary report respectively than to update the initial joint programme theory which was shown in Figure 3.

The feasibility study also found that the pre-consultation questionnaire was useful to patients and GPs and took very little of their time. Recruitment rates (recruits per SMS invitations sent) was 26%, which was higher than the 15% target. However, the technical process surrounding the pre-consultation

form required too much support from the research team to be easily rolled out³⁴. The realist evaluation should be read in this context.

3.2.3 Revised programme theory

Our analysis of the data from 63 interviews enabled us to revise the programme theory for the pre-consultation form. This is shown in Figure 14. This presents fourteen context-mechanism-outcome configurations (CMOCs) identified in the data in a single diagram showing interlinked context (what works), mechanisms and outcomes. "What works" can be understood as how the interventional subcomponent needs to be implemented (to alter context) to achieve the outcomes. The mechanisms are processes that are 'triggered' by the context to cause outcomes. Some mechanisms are only activated for certain types of patients and consultations. This information on which types of patients and consultations are shown by the numbers in brackets in the green mechanism boxes. Two of the outcomes are more distal than the others and for these, the context in which they are achieved is represented by another outcome in the programme theory (functioning as a context for that CMOC).)

3.2.4 Context (what works, for whom and what circumstances)

This section focuses on providing an overview of the context through which the mechanisms are activated using a realist evaluation framework of "what works, for whom, in what Table 6. Patient and practice interviewees for the pre-consultation form.

	Intervention Development	Feasibility Study
Patients		
Site 1	6	6
Site 2	2	7
Site 3		9
Site 4		8
Site 5		
Site 6	4	
GPs		
Site 1	2	2
Site 2	2	2
Site 3		2
Site 4		1
Site 5		1
Site 6	2	1
Administr	ators	
Site 1		1
Site 2		1
Site 3		1
Site 4		1
Site 5		1
Site 6		1
Total	18	45

*Sites 1, 2 and 6 were in the Intervention Development Study as well as Feasibility. Sites 1 to 4 were intervention sites and sites 5 and 6 were control sites.

circumstances". Details about the CMOCs in which these contexts function may be found in below in section 3.2.5.

3.2.4.1 What works

Key elements of the pre-consultation form that worked well were: 1) the simple format of the form combined with the ability to add more information where required 2) patients taking time completing the form accurately and with the right amount of detail 3) the colour-coded format of the report 4) GPs ensuring that they read the report properly 5) GPs letting the patients know they have read the form and then listening.

The form was most useful for GPs when patients took time to complete the form accurately and provided an appropriate amount of detail, providing free text for an item if they scored the middle option or lower. This was particularly the case when it was a health concern or a need for information or support.

GPs found the colour-coded fixed format traffic light system worked well as it made it easy for them to quickly read and assimilate the information.

It's certainly much quicker for a clinician to look at than an [practice electronic triage system] is. You just really want your eyes to go straight to, what it is that matters, and the traffic light system enables you to do that and getting engaged. (GP 1, Feasibility Study)

The colour-coded format was occasionally mis-leading, when the patient had a chronic condition which involved chronic pain, physical symptoms and/or anxiety, but that was not necessarily what they wanted to talk about that day. GPs normally established this quickly at the start of the consultation. Many patients tend to select the "slightly" option for mild ongoing problems that they don't need to discuss with their GP and if this was selected but the patient had not put any more detail, it worked best if the GP did not raise that in the consultation as an issue.

GPs letting patients know that they have read the form at the start of the consultation was very important to patients, and a driver for patient perceiving that the GP was prepared and feeling listened to and taken seriously. One patient contrasted this approach with the previous approach of the GP greeting the patient and asking how they can help.

I felt like I wasn't going in blind because normally, you know, I remember before COVID you'd walk in and GP would look at you like, okay, like why are you here? Even though you've spilled your life out to the receptionist? [laughs] They still haven't got a clue while you're there.... So, you think, why were you asked all these bloody questions when either it's not been relayed or they've not read the notes. (Patient 27, Feasibility Study)

One GP noted that she had also experienced this reaction from patients.

Well I think it's just that acknowledgement, because when you say... normally in the consultation I say, 'Hello, it's Dr Jones returning your call, how can I help today? And sometimes they'll be a bit grumpy, they'll say, 'Well, I told your receptionist.'[...] or they'll say, 'Well, you've got my notes in front of you haven't you?' [Laughs] so I think this just was a way of acknowledging to patients that you were interested in them, and that you had read what they had written. (GP 5, intervention development study round 3)

Although the majority of patients said the GP made it clear they had read the form, one or two did not think it had been read. Of the patients who thought the form made no difference to their consultation two of these were unsure that the GP



- 4. Patients who are feeling overwhelmed or down about their health
- 5. Patients who feel their problems have not been previously addressed or taken seriously
- 9. Where health concerns or low mood or life impacts are raised
- 10. When patients have a problem that requires reference to the medical notes, the BNF formulary or patient pathways.

******Some evidence for this, but less than for other outcomes

* Substantial evidence for this, but also evidence for divergent cases – where the opposite may have occurred

Figure 14. Pre-consultation form, revised programme theory.

had read the form. This underlines the importance of the GP making it clear that they have read the form, by reflecting back the patient's words to them, or bringing up issues before the patient does.

3.2.4.2 For what type of patient

The pre-consultation form works best for: 1) Patients who find GP consultations difficult 2) patients who are comfortable with technology 3) patients who have a concern they find difficult to voice 4) patients who feel overwhelmed or down about their health 5) Patients who feel their problems have not been previously addressed or taken seriously. The last three of these are specific to particular mechanisms and are covered in 4.3.3 (CMOCs). The first two (patients who find GP consultations difficult and who are comfortable with technology) apply to the intervention generally. One patient described how she felt about GP consultations:

Because when you go to the doctors it's really quite nerve-racking, a nerve-racking experience. So actually being able to write something down and have a little bit of time to think about it, like your symptoms and what you're feeling and everything else was really helpful to me. (Patient 28, Feasibility Study)

Many of the patients who were interviewed commented that they were comfortable with technology:

I'm quite used to sending things, bank details and all kinds of stuff by my phone [...] I'm aware some people might not, but I do feel fairly secure, rightly or wrongly, with that kind of thing these days. So it didn't really, I have to say, bother me too much. I wasn't worried about it. (Patient 23, Feasibility Study)

We were unable to interview non-responders, so do not have information on what types of patient did not complete the form, but practice administrators were able to see the age profile of the patients who they messaged and had visibility of who responded and some of these felt younger people responded.

3.2.4.3 In what types of consultations

The pre-consultation form was most useful for consultations 1) where new information was raised, 2) where complex, multiple or layered problems and situations were discussed, 3) which were conducted by telephone 4) where information about health concerns, low mood or impacts on life were raised, or 5) where the problem required reference to the BNF (British National Formulary), funding rules or the patient history. The last two of these are specific to particular mechanisms and are covered in 4.3.3 (CMOCs). The first three (new information, complex problems and telephone consultations) apply to most of the CMOCs and are discussed below.

GPs felt the form was most useful for new problems. The practice appointment booking procedure followed by Site 1 meant that a disproportionately high number of patients who had follow-up appointments booked were sent the form. This practice had also been involved in the Intervention Development study before the practice booking policy changed. The GP commented:

Last time [Intervention Development], we had a lot of new patients, and it worked really well, because they were not such follow up things. Logistically, it was fine [this time]. But just from a quality of information, it probably wasn't as useful, I don't think. (GP 2, Feasibility Study)

Some patients agreed with this:

For me, particularly, I don't think it made any difference to the way the GP listened. It was pretty much the same as my normal consultations, but I do think if it was a new problem, then it would be more helpful. (Patient 11, Intervention Development Study round 3)

This patient was consulting about a problem they had seen the GP about several times before, and didn't have any new information to raise, so found the form less useful.

Patients who had complex problems, multiple problems or problems with different aspects to them found the form very useful. Few people with single simple problems completed the form, but when they did, they found the form less useful:

They [the questions on the pre-consultation form] weren't relevant to me because I mean to be fair all I was asking was a very simple question. Because it's one of those things where literally a two-minute conversation was all I needed. (Patient 10, Intervention Development Study round 3)

Some GPs, on the other hand, still found the forms for these very simple problems useful, because they highlighted that there was no hidden agenda to be concerned about and sped up the consultation (see CMOCs). Figure 15 shows an example of a report where the patient found the consultation useful, next to one where the patient did not find it useful. As the figure shows, the most useful consultation was one where multiple aspects of the problem were raised. The least useful was for a very simple problem.

Some patients found the pre-consultation form more useful over the phone because communication is harder than in a face-to-face consultation so the form acts as an aid to communication.

Definitely, definitely, [more useful over phone] because obviously the phone, you've got the delays and we were both using ... well I was on a mobile phone when she contacted me - there's always that sort of delay in speech. And I felt that she was able to read what I'd put prior to actually speaking to me, so she understood better. So I felt from the telephone interview point of view the form worked very well. (Patient 2, intervention development study round 1)

This patient felt that the GP understood better because she had read the report, and therefore the delays in the phone were less disruptive that they would have been otherwise. Some GPs and patients, however, thought it made no difference whether the form was used over the phone or face-to-face.

3.2.5 Context mechanism outcome configurations (CMOCs)

In this section, more details are provided on the fourteen CMOCs shown in the programme theory (Figure 13).

Outcome: Issues raised that might not have been raised otherwise

CMOC 1: Invitation to raise concern->Issues raised that might not have been

When patients who have a concern they find difficult to voice receive a clear user-friendly form with specific questions about their problems (C), issues are sometimes discussed that might not have been otherwise (O) because the patient feels invited to raise their concern (M). (Box 1)



Report for 9-Pat-Site 2-28072021

Your patient gave the following reasons for attending:

1 I have been struggling over the past year and I feel it's down to my hormones

consulted about 1-2 times

The	The list below highlights some other problems your patient might have.					
1						
1						
2	Other Physical Symptoms	(1)	4		Quite a bit	Headaches Cystitis Waking up through the night Tired Hot sweats Low and feeling sad all the time Memory loss Generally not feeling like myself
3	Effect on normal activities	(1)	3	۲	Moderate	I am busy a lot of the time, so it's exhausting to 'stay normal' and keep everything going
4	4 Low mood / Anxiety (2) 5 🐼 Extreme Feel like crying a lot of the time					
5	Concerns about serious illess	(1)	2	٢	Slight	Slightly worried symptoms might indicate a serious illness.

Report for 5-Pat-Site 1-20072021

Your patient gave the following reasons for attending:

1 Results of blood test					consulted about 3-4 times
2 Allergy test					new problem
The list below highlights some	other	prot	olems you	ır patient mig	ht have.
Green indicates no problems.	Light g	reen	indicates	slight proble	ems. Amber is moderate and red is severe.
1 Pain	(1)	1	٢	None	
2 Other Physical Symptoms	(1)	1	٢	None	
3 Effect on normal activities	(1)	1	٢	None	
4 Low mood / Anxiety	(2)	1	٢	None	
5 Concerns about serious illes	; (1)	1	٢	None	

Figure 15. Excerpt of pre-consultation report for different types of problems.

Because they were explicitly asked on the form, patients who had concerns they found it difficult to voice found it easier to raise issues, including low mood, health concerns, life effects and support needs. One patient, who was expert in her own condition (lupus) had been worried the pain in her side might be liver related, but hesitant about raising this for fear she "looks like you're trying to tell them [the GP] what they should know":

So with lupus [...] you're always second guessing what might be going on underneath. [...] Sometimes you don't want to say that necessarily outright [...] with doctors, you think, oh, if it looks like you're trying to tell them what they should know. I know enough to know that lupus affects the liver. I know enough to know my blood test shows that I've got abnormality in my liver. So in the back of my head, there's almost a little bit is this discomfort I'm having for the last six months that I've not mentioned to anybody, is this a reason why I've now got this abnormal blood test in my liver? [...Now] because I've written it on the form, she could actually say, 'Oh, I know you mentioned it's possibly the liver. Actually, it'd be extremely rare for anybody to actually have pain in the liver like this. It would be very, very rare. So I think we need to keep an eye on it. If it keeps getting worse, then we'll need to have a scan' [...] So it made a huge, huge difference, so yeah. (Patient 23, Feasibility Study)

This patient had a health concern which for six months she had avoided explicitly asking the GP about. The invitation to share health concerns on the form meant this was raised and addressed in the consultation. Another patient who attending with a skin condition and low energy levels similarly felt that, because she was invited to include it on the form, the GP picked up on her anxiety where she might not have done previously.

R: *I* think that she understood it was troubling me more than just a visual issue. I came out feeling a little bit more that I was gonna be looked after (Patient 24, Feasibility)

I: Do you think that was as a result of the questionnaire that you filled out? Did that give her information that she was looking at?

R: Yes, because the questionnaire asked about anxiety and stuff along those lines, and that's, obviously, where she may have picked that up from.

GPs also noted that there were things raised on the form which may not have been had the patient not been invited to share them:

So things around sort of suicidal planning [...]; I don't know whether they would have told me those things in the consultation or not. So it maybe that those things wouldn't have come out if they hadn't written them down [...] it definitely did sort of ring alarm bells that hadn't been there when I spoke to the patient previously I suppose. (GP 7, Feasibility Study)

This GP was referred to a patient who had disclosed significant suicidal planning. The GP said this level of planning had not been evident when he last spoke to the patient, and he was not sure if the patient would have disclosed all the details without being asked to give more information on the form.

CMOC 2: Greater patient reflection->Issues discussed that might not have been otherwise

If a patient has multiple, layered or complex problems and this patient takes time to complete the form accurately (C), issues can be discussed that might not have been otherwise (O) because the patient reflects more on what their concerns and priorities are in advance (M). (Box 2)



Some patients felt like the action of completing the form made them reflect on why they wanted to see the GP and prepare better for the consultation. Patients explained this as follows:

I had multiple things that I wanted to talk to her about, multiple symptoms, and often you can forget whereas it made me actually think, 'well, actually I want to talk about this. 'I want to talk about that', so it made me better prepared. (Patient 12, Intervention development study round 3) It makes you think carefully about why you know why you are bothering the doctor if you like or yeah it focusses your mind on your problems. (Patient 33, Feasibility Study)

The first patient above had multiple symptoms and the action of the form helped her reflect on what she wanted to discuss and prepare what to say. The other patients had a problem with different layers to it; pain that was affecting her sleep, quality time with her family, was causing low mood and she was also worried about the underlying cause.

Some GPs also felt that patients were more prepared in the consultation. One GP described her patients as more "focussed". This GP acknowledged that this was "my perception of their perception"; i.e. the GP could not say categorically whether her patients were more prepared, only how prepared the patient seemed to her.

There were also a substantial number of patients who thought completing the form did not necessarily make them better prepared, because they tended to prepare for a GP consultation naturally anyway, and the form appealed to them because it was a mechanism for doing that in a way that could be shared with the GP.

No, I didn't [feel the form made me more prepared] if I'm honest [...] because on those occasions when I do see my GP, I always make a note anyway as I would if I was writing a letter to the bank or anything. I'm a writing down sort of person, so I'd make a few notes anyway, so that's what I do and in all honesty, that's more about the kind of person I am. (Patient 40, Feasibility Study)

CMOC 3: GP can raise problem before patient->Issues discussed that might not have been otherwise

If a patient has a concern they find it difficult to voice, low mood or a health concern, or multiple, layered or complex problems and this patient completes the form accurately, and this accurate colour-coded report is shared with the GP (C), issues can be discussed that might not have been discussed otherwise (O) because GPs have a written record to refer to in the consultation so they can raise the concerns themselves if the patient forgets (M). (Box 3)

Box 3. CMOC 3 diagram				
Patient has concern difficult to voice, low mood, health concern or complex / layered problem and				
GP reviews accurately completed form Context	GP can raise problem before the patient Mechanism	Issues discussed that might not have been otherwise Outcome		

Some patients liked having the written record because GPs could bring up problems before the patient had to. One patient explained that she was worried her symptoms might be cancer and the form helped her express this:

I think everyone's been in the situation where they go to a doctor to talk about something that they find hard to talk about or they might find it difficult to voice their concerns. [...] in the form I filled in this time it said what are you really worried about, and I was able to write down the word cancer because that's something that worries me, worried that I won't be here for my children, if I had to say that to the doctor I'm sure I would've just burst into tears saying it and then I wouldn't have been able to have a particularly productive consultation. But because I was able to write it down before and she knew what was on my mind and I knew she knew what was on my mind, she took the consultation really seriously and was very clear about the steps she was going to take. (Patient 38, Feasibility Study)

This patient found it difficult to voice her concern but the form invited her to share what her concern was. This then led to this issue being addressed in the consultation. The GP took her concerns seriously and put together a plan of action, which included a fast-track cancer referral. Another patient who was attending about an ongoing cough used the form to raise the fact that he was very concerned about his weight and was relieved when the GP raised this.

R: *The doctor picked up on something on the form which we then spoke about (Patient 21, Feasibility)*

I: What was the issue that she picked up on?

R: My weight. It's is a massive issue, and it's so hard, and she agreed. I don't know whether if it was the way I'd written 'cause I said something like, 'I need to lose five stone' – I can't see myself losing more than two ever, and that's at a push, and that's like a mountain. That was a bit of a cry for help – I suppose, subconsciously, it probably is, actually.

GPs also commented that it was useful to be able to raise concerns before patients did, in particular for sensitive problems. One GP gave the example of sexual health:

there was one around sexual function, potentially, that is quite a difficult thing for someone to start talking about in a consultation. Probably would have got there in the end, but it just allowed that to come out much more easily, and without as much difficulty or embarrassment on anyone's part, I think. So, I think it is really useful for those kinds of things, that it's something that's slightly sensitive to bring up. (GP 2, Feasibility Study)

Outcome: Time used more effectively

CMOC4. Dialogue begins before consultation->time used more effectively

When patients complete the form accurately and this accurate colour-coded report is shared with the GP (C), time can be used more effectively (O) because the dialogue begins before the consultation (O) (Box 4)



Some patients commented that the GP reading the form made the consultation more efficient because the dialogue had already started:

by completing the form first you feel as if... and the doctor acknowledging that they've read it, I think that you're able to make probably more effective use of the time because the dialogue has already begun. (Patient 38, Feasibility Study)

These patients felt that completing the form enabled time to be used more effectively in the consultation. Some patients felt that because the GP knew what their problems were in advance, this helped speed the consultation up and save time for the GP. Other patients felt that the consultation was not necessarily shorter, but the time was used more efficiently.

GPs felt that questions about mental health and life support were often reached more quickly when these were completed in the form.

[It was efficient because] you weren't having to ask those questions about, how is life, have you got enough support, how's your mental health [...] sometimes when you're asking patients about those things they'll go into a huge 10-minute monologue about how they're struggling, and actually [laughs] you've kind of got that little precis of how they're managing. It's quite efficient I think. (GP 5, Intervention development study round 3)

The GP felt that, without the form, she would have had to ask her patients more questions about the impact of their conditions on their lives and their responses would have been much lengthier than was shown on the form.

The context of the form being accurately completed is key to the effective use of time. There were exceptions to the form saving time when patients with ongoing issues may have flagged anxiety or chronic pain on the form, but that wasn't necessarily the issue that they wanted to talk about that day. In these cases, if the GP raised the issue, then this took more time.

CMOC 5. *GP* address what is important to patients-> time used more effectively

When patients complete the form accurately, and this accurate colour-coded report is shared with the GP (C), consultations time can be used more effectively (O) because GPs can spend the consultation time addressing what is important to the patient (M). (Box 5)



The form enabled GPs to quickly focus on what mattered to the patient, and this meant time was used more effectively. This applied to patients with very straightforward problems as well as complex problems because the GP could focus on that straightforward problem without concerning themselves about a hidden agenda. One patient who was attending because of rectal incontinence and mucus felt the form was useful because it asked about her anxiety and health concerns. The patient was, in fact, not particularly anxious about the problem, but wanted to resolve the symptoms.

we ruled out are you worried by what you are there for? No, I wasn't, I was just trying to resolve an embarrassing situation, and that question was kind of covered in a way. Is it causing me a mental issue? No, it's not. There were certain things there that were ruled out because I answered it honestly. (Patient 34, Feasibility Study)

Even though this patient did not have any health concerns or anxiety about the problem, she felt it was useful to be asked the question, as she was aware that the GP might have anticipated her having anxiety or concerns about her problem. She pointed out that her accurate responses to the questions were important in making this work.

When complex or multiple issues were raised, being able to focus on what was important to the patient also meant that time could be used more efficiently:

I think it worked well in terms of sort of getting the relevant information and sort of getting to the heart of the issue earlier on that you probably would have done otherwise in the consultation I think. (GP 7, Feasibility Study)

This GP had anticipated, on doing the training, that there would not be time to deal with the extra issues raised in consultations as a result of the form, but instead found that she was able to "get to the heart of the issue" in terms of what was important to the patient earlier in the consultation and therefore use time more efficiently.

There were some exceptions to this, where the form did not help the GP to focus on the patient priorities. This was when the patient had ongoing chronic problems and completed the detail on the form based on those chronic problems but they did not want to discuss those problems on that day.

CMOC 6. GP prepare in advance-> time used more effectively

When patients with a new problem or a problem that requires reference to their medical notes, the BNF formulary or referral/funding rules complete the form accurately, and this accurate colour-coded report is shared with the GP (C), consultations can be more time efficient (O) because the GP can prepare for the consultation in advance (M) (Box 6)



Some patients felt that the GPs were more prepared for the consultation and this meant consultation time was used more effectively. This was particularly when a patient had a problem that required the GP to refer to their medical record, the BNF or funding/referral rules.

I felt confident that the GP had read the information and was prepared for what I wanted to talk to her about and [...] I felt that she had already looked up some test results that I had had previously and was able to comment on those straightaway. (Patient 12, intervention development study, round 3)

This patient felt that, by reading her form and understanding why she was attending (fatigue, headache and joint pain) the GP was able to look up previous test results, which made the consultation more efficient. GPs also noted that there were times when they were able to be better prepared for the patient by looking up relevant information in advance.

if you know what they're asking, particularly those more complicated things around funding rules [patient wanted fertility checks], or those kinds of things, you can look them up beforehand, which is really useful. Normally, I'd have been looking it up afterwards and then saying, 'Actually, we need to this first,' or, 'We need to do that first.' So, it is really useful for those more complicated things where the rules change all the time, but it's difficult to remember. (GP 2, Feasibility Study)

I was able to prepare. And, looking through their medical records to see any history that I could relate to what they were talking about, whether it was a new. So it allowed me to almost have a mini-plan depending on what the patients would tell me. (GP 4, Feasibility Study)

GP 2 explained that, knowing the patient was going to request a fertility check, she was able to look up the rules on this in advance, in the context of how the patient's life was affected, where previously the GP would have had to look up these things afterwards.

Outcome: Wider range of support offered tailored to the patient's needs

CMOC 7. *GPs* address what is important to the patient->wider range of support offered tailored to the patient needs

When patients complete the form accurately, and this accurate colour-coded report is shared with the GP (C), the GP can offer a wider range of support which is tailored to the patient needs (O) because GPs can spend the consultation time addressing what is important to the patient (M). (Box 7)

Box 7. CMOC 7 dia	ıgram	
Accurate colour-coded report shared with GP	GPs can focus on what is important to the patient	Wider range support offered tailored to patient needs
Context	Mechanism	Outcome

Some patients felt the form enabled the GP to quickly focus on what was important to them, enabling them to deal with multiple issues, where previously there may have been only time for one.

I think I raised a lot of things on the form. The talk with the doctor was pretty much jam packed. There was no waffling or dallying about. That's what I mean; it was just very productive because there were a few things on the form. (Patient 44, Feasibility Study)

This patient went on to describe a targeted conversation, focussed on her needs which resulted in a referral to physiotherapy.

Some GPs also thought the form often enabled them to quickly focus on what was important to patients and this meant that they could offer the patient more support, either within the consultation, by a treatment plan, referral or signposting.

One GP explained how this was particularly useful for a patient with an anxiety or a health concern, because the GPs priority is often to focus on a diagnosis and the immediate safety of the patient. However the patient's priority may be to deal with the anxiety, and this can often be done quickly without compromising the diagnosis or patient safety. Some GPs also felt that they offered a broader, more holistic range of support to patients, because they were more aware of the impacts that their problems were having on their life.

say for example someone has got pain, normally you'd be focussing on their pain and talking about painkillers, but you wouldn't have acknowledged, 'Actually I can see you're really struggling with this pain, I would like to support you.' So that was the kind of thing that I was saying at the beginning of the consultation, you'd say that and then you'd say, 'Well tell me about the pain,' and go through the normal management of their pain, but then you would say, 'Well actually let's think about support we can give you,' be that mental health support or social prescribing, or follow up consultation. Yes, I think it was good. (GP5, Intervention development study, round 3)

This GP felt that whereas previously she would have focussed on pain and medication, she now focussed on other support that could be offered. The patient in question, who had been consulting with back pain, also agreed with this assessment:

You're not [normally] asked that when you go to the doctors. They don't say, 'how does this make you feel?' It's kind of, 'alright here you go' or 'we'll see you in a couple of weeks'. I think that made a massive difference because now I've got more help with how I'm actually feeling in myself [...] just by simply clicking on a link and answering questions. It has really benefitted me. I would have gone to my appointment, had my back checked over and walked out but now I've had several things for me that have really benefited me. (Patient 9, Intervention development study round 3, patient of GP 5 quoted above)

This patient felt that because she was explicitly asked about the impact of her back pain on her life the GP was able to address those impacts in terms of offering more support.

CMOC 8. Written record emphasises importance->wider range of support offered tailored to the patient's needs

When patients who feel their problems have not been previously addressed or taken seriously complete the form accurately, and this accurate colour-coded report is shared with the GP (C), a wider range of support is offered tailored to the patient needs(O) because the written record emphasises the importance to the GP (M). (Box 8)



Patients felt the written record was important, not only because it was a reminder for GP so that their GP could raise the issue before the patient did (CMOC3), but also because the written record emphasised the important to the GP and may therefore have prompted action:

And also, when you're writing it all down, all your symptoms, you feel more in control, because you're actually writing it down at the end of the day, rather than going into the doctors and saying it, you're writing it down. The action of writing it all down makes it more real to the doctor and to you because you've written it down there's a record there. (Patient 28, Feasibility Study)

I felt like they almost had a different tone with me [because of the pre-consultation form] Well more than normal because a lot of the stuff you can't really see it on me. It's not that they disbelieve me normally but it's just that I suppose they are being accountable aren't they because it's in writing. It's like this needs to be dealt with or this needs to be talked about. (Patient 44, Feasibility Study)

For these patients, the written record was more real to GPs than a verbal report. They felt the GP was more accountable because of the written record and they were more likely to be taken seriously.

Some patients believed that action was taken because GPs took the written record seriously. Patient 2 believed the pre-consultation form helped move her consultation forward, and contributed to the GPs decision to refer her to a physiotherapist. When asked how she thought the form made a difference she said:

because I think she could see the words, and I think sometimes words can have a bit more impact written down than sitting face to face. (Patient 2, Intervention Development Study, round 1) Patient 2 believed that having the written record emphasised the importance for the GP, increased her accountability and prompted her to make a physiotherapy referral.

GPs did not raise this in earlier interview rounds. Because patients raised it, some GPs in later interviews were explicitly asked, and most disagreed with the patient's assessment:

Interesting they feel that. I don't think the doctor feels like it's more real. [laugh] (GP 3, Feasibility Study, GP of patient 28 quoted above)

One GP did agree that he might take the written record more seriously than the verbal account. This GP said that he felt many of his patients often exaggerated their symptoms.

I would probably say I thought about that less [patient is exaggerating] when I was reading the reports. Maybe them putting it down on paper gives it some kind of more... more believable or more legitimate [...] it's an extra step there that takes an extra bit of thought for the patient and maybe that extra thought is what makes it more legitimate, more real. (GP 9, Feasibility Study, GP of patient 44 quoted above)

This GP felt that the extra step of having to physically write the symptoms down may cause the patient to reflect more and be less likely to exaggerate, so he would take what was written more seriously than what was spoken. GPs acknowledged that the form sometimes highlighted the number of times a patient had consulted about a problem before, and this sometimes made them take the patient's problem more seriously:

one thing I did think about this form which is actually in blue was quite useful to know, was that she'd said she'd had nine episodes of depression in 16 months and that was quite useful to know. I perhaps might not have known how frequently she was feeling she was becoming unwell and not functioning [...] and not been able to work for a year is quite a big thing isn't it really. That's kind of a big red flag if they're not getting back to work really. So yes, that was quite a useful form especially because I didn't know this person very well so that was, you know, I do now and yes I've had quite a few more consultations subsequently with this person so. (GP 3, Feasibility Study)

As this GP explained the patient's multiple depressive episodes was new information for her. However, the patient may have thought that, as this should have been available from her medical record, the GP was taking it more seriously because it was in writing. Another GP said that if the top of the form indicated that the patient had consulted about this problem multiple times, it made her focus more on making active progress towards a resolution within that consultation, which may have created a similar impression among patients.

Outcome: Patient more satisfied with the consultation

CMOC 9. GP Listening and acknowledging->patient more satisfied

When GPs read the form, let the patient know they have read it, reflect the problems back to the patient and then listen carefully listening carefully (C) patients are more satisfied with the consultation (O) because they feel listened to and taken seriously (M) (particularly when they feel they have not been listened to or taken seriously in the past) (Box 9)

Box 9. CMOC 9 diagram						
GP makes it clear form has been read, asks if there is any other problem and listens without interruption <i>Context</i>	Patient feels listened to and taken seriously Mechanism	Patient more satisfied with the consultation				

A lot of patients said they felt listened to and taken seriously when GPs acknowledged that they had read the form. This was a key driver of patient satisfaction. Patients who completed the form really appreciated the GP acknowledging that they had read the form immediately. Because GPs were not used to this way of opening a consultation it initially felt awkward to some of them; they were concerned it might interfere with the consultation and might seem like they were listening less than usual. However, most patients thought the opposite – they felt more listened to than usual.

R: She did listen really well. Yeah, no, I really felt listened to and that does definitely not always happen, so I did. I did come off the phone and I said to my husband, I just said, 'That was just so good. I actually felt really listened to', (Patient 23, Feasibility Study)

I: You feel the questionnaire was a part of that?

R: Yeah, I do. I do.

This patient attributed the GP listening well to her completing the form in advance. Some patients said it was hard to tell if this was as a result of completing the questionnaire, particularly if it was their first time seeing that GP. One patient who tended to see the same GP explained that she already held this GP in high regard, yet felt more listened to than usual:

I've seen her [the GP] previously a couple of times before this questionnaire. When I received the link, I'd seen her two times prior. It was a completely different experience. I was definitely listened to a lot more. Don't get me wrong, she's a wonderful doctor and she listens anyway but this time I felt more listened to. (Patient 9, intervention development round 3, patient of GP 6 quoted below)

In many cases, the GPs seemed to underestimate the impact of them acknowledging they had read the questionnaire and then listening had in terms of patient satisfaction. When this patient's GP was asked about whether she thought she listened more she said:

I don't think I was listening more than I would have done, but that's probably just acknowledging more, and actually actively talking back to them what they've said. I think just that summary of saying, 'From what you've written I can see that x, y, and z. That's how we train our students to reflect back to patients, but perhaps I don't do that enough and it's making sure that you do that, which is quite nice. (GP 6, intervention development round 3, GP of patient 9 quoted below)

The GPs explained that she was not listening any more than usual, but that the process of reading the pre-consultation form and reflecting it back to patients probably gave her patients the impression she was listening more. As the GP pointed out, this is not a new technique; these are core consultation skills which are taught to medical students, but under the pressures of a normal daily surgery it is easy to forget the importance of this, and the COAC intervention provided a framework for doing it early in the consultation so the patient felt listened to and validated.

CMOC 10. *GP* gauging patient mood and showing empathy->patient more satisfied

When patients who are feeling overwhelmed or down about their health problems complete the form accurately and GPs read the form patients (C) are more satisfied (O) with the consultation because the GP can gauge the patient's mood early and show appropriate empathy (M). (Box 10)



Some patients were satisfied with their consultation partly because the GP treated them with empathy and kindness. These were patient who were feeling low or overwhelmed, and some of these felt the form being shared with the GP was what prompted the GPs manner:

her knowing it already helped... helped me, especially because she knew I was feeling quite stressed and low and anxious over everything. And knowing that already, she could be a little bit more gentle with me, if you like... a bit more mindful of that fact that I was feeling quite low, if you like [...] it was actually quite refreshing to have someone be a bit sympathetic. [Patient 7, intervention development study, round 2)

This patient felt the GP was more "gentle" with her, because she knew from reading the form how the patient was feeling. This made the patient more satisfied with the consultation, referring to the experience as "refreshing".

CMOC 11. Issued addressed and more efficient and supportive consultations->patient more satisfied

When all their issues are picked up, when the consultation seems efficient and when a wider range of support is offered (Outcomes -> Contexts), patients are more satisfied with the consultation (O), because their needs and expectations have been met (M). (Box 11)



The majority of patients interviewed felt more satisfied with the consultation than usual. Although this was partly because they felt listened to and taken seriously (CMOC 9), patients also felt more satisfied when all their issues were picked up on and they were offered support for them. One patient had completed the form describing headaches, low mood, memory loss and other symptoms which she thought was related to the menopause. She expressed described her satisfaction with the consultation as follows:

That really was easy for me [completing the form]. It was fine. There was nothing that I would have changed. And obviously I got what I wanted at the end. I knew what was wrong. I knew how it should be dealt with, and I got my end result. So, everyone's a winner really. (Patient 29, Feasibility)

There was evidence from the interview that the efficiency of the consultation partly led to this sense of satisfaction that the patient described:

I felt like she'd read everything I put, and then it was just really easy and quick. (Patient 29, Feasibility)

The patient described it as "easy and quick" because the GP had read the information in advance. The patient also felt that all her issues were addressed.

I felt like it was, I said because I had the time and space, I said everything that I wanted to say. That there was nothing that I just thought, shit, I should have said that I should have said this, because I'd already said it. (Patient 29, Feasibility)

Finally the patient felt that the GP was able to provide more support because of the form:

She read it and made a sensible decision as to how to move forward. (Patient 29, Feasibility)

In the quote above, patients 29 explains that the GP was not only efficient, and uncovered her concerns, but also provided her with support and a plan on how to "move forward". This patient therefore provides evidence that these three outcomes led to her satisfaction. The consultation was a telephone consultation and she felt that the pre-consultation form worked particularly well for this.

Outcome: Improved patient wellbeing

CMOC 12. Wider range of tailored support offered->improved wellbeing

When GPs offer a wider range of support tailored to the needs of the patient (Outcomes -> Contexts), the patient's wellbeing,

health, knowledge or empowerment can improve (O) because they are provided with the means to address their needs (M) (Box 12)

Box 12. CMOC 12 diag	ram	
Wider range of support offered tailored to patient needs	Patients are provided with a means to meet their needs	Improved wellbeing, health knowledge and empowerment
Outcome functioning as context	Mechanism	Outcome

Some patients described how being offered more support improved their well-being, health knowledge and empowerment. Improved wellbeing, health knowledge and empowerment is the most distal outcome shown in the revised programme theory, as often patients have to go through with the treatment plan before their health improves so there is less evidence for this outcome than others. However, there is still some evidence in a small number of patients.

One patient who put "follow-up" as her reason for appointment was there to discuss indigestion and chest pain. She had previously been prescribed Gaviscon, and had a follow-up consultation to check the problem was resolved. Because she was specifically asked on the form, she disclosed that she was struggling to lose weight, and would like some counselling support. The patient described her outcomes:

Well I was really pleased because, as I say, he referred me, so I'm going to join Weight Watchers, and also I've booked an assessment to maybe get some counselling. And also he reassured me about the indigestion. So I felt that it was a really useful conversation [...] I think it [the questionnaire] definitely did help, because I probably wouldn't have mentioned the other things, we'd have just talked about the indigestion and that would have been it. (Patient 41, Feasibility Study)

The patient said she would not have raised the other issues had she not been asked in the form, so that additional support she received with lifestyle and mood was as a direct result of completing the form. This patient's GP was also interviewed and mentioned this patient:

you've gotta remember this is also kind of like not face to face as well so this is where this actually proved quite helpful because [...] you know to broach someone's weight is a bit more difficult over the phone may be than face to face [...] And so a five minute telephone call could have very easily been well '[how] is your indigestion', 'oh it's a bit better with the over counter stuff', 'right fine'. But this gave the option to explore, rather than terminating the call quite quickly, to actually really look into what other things she thinks would be helpful and hence the discussion about her weight which she made the focus of the consultation a lot more. [laugh] you know it's made the consultation longer but the patient got something more out of it as a consequence. (GP 8, Feasibility Study)

The GP agreed that the problem probably would not have been raised had the patient not put it on the form. He pointed out that the consultation was not more time-efficient, and in fact took longer, but was more valuable to the patient.

Another patient, who was very positive about the contribution of the pre-consultation form to her consultation outcome was asked how she thought her outcome was improved through completing the form:

I think an example is that lots of other antidepressants lead to weight gain quite a lot, and in the past a lot of doctors have said that it was just my hormones and things like that. But being able to write down on the form that I wasn't happy with the weight gain, I was able to raise that. I probably wouldn't have raised that before if it was face-to-face, I would have felt they would say something similar about my age, and this, that, and the other [...] So, I changed my medication, and I also mentioned on the form -I was able to do so with the multiple boxes - that I had trouble sleeping. So that was taken into consideration, and now I sleep much better, which means I'm generally a happier person! [..] So, I think yeah, it's had a knock-on effect definitely, because I feel I've had a tailored outcome instead of if I just walked in and tell him I'm struggling with my mental health, he may have just put me on the first medication, the most common one. Even though that's not always the best individually; I was able to have a more tailored consultation. (Patient 47, feasibility Study)

This patient strongly felt completion of the form led to improved sleep and changed medication, which she hoped would help with weight loss. The patient wrote on her form that she was unhappy with the side-effects of her medication, that she had raised this with multiple GPs and felt that poor continuity of care was hampering her progress. The patient felt that she would not have been able to explain this so succinctly verbally, and that by being able to read her history and concerns on a single form, the GP took action which resulted in immediate improvements to her well-being.

CMOC 13. Listening and acknowledging->improved wellbeing

When GPs read the form, let the patient know they have read it and reflect the problems back to the patient before listening without interruption (C), this can improve the patient's wellbeing, health knowledge and sense of support (O) because they feel listened to and taken seriously (M) (which is therapeutic in itself). (Box 13)



Some patients felt that being listened to and taken seriously was therapeutic in itself, and improved their well-being, often their mood. One patient explained:

I actually felt really listened to', because quite often, especially with (disease), I can feel completely on my Page 27 of 37 I was more reassured because I felt like I was actually being taken seriously and listened to. (Patient 29, Feasibility Study)

Patient 23 explained that being listened to made a "huge difference" not just in terms of her satisfaction with the consultation, but with her sense of support going forward. She no longer feels on her own with her illness. Patient 29 described herself as feeling reassured because she was listened to. As with the other CMOC related to improved well-being, there was less evidence for this than the other outcomes.

Outcome: Patient confidence in seeking healthcare

CMOC 14. GP Listening and acknowledging->patient confident in seeking healthcare

When patients feel their problems have not previously been addressed or taken seriously, and GPs read the form, let the patient know they have read it and reflect the problems back to the patient and then listen carefully (C), patients are more confident in seeking healthcare in the future (O) because they feel they have been listened to and taken seriously this time (M). (Box 14)



Some patients felt that, because they felt more listened to, they were more confident in seeking healthcare from their GP in the future. This was particularly the case for patients who did not feel they had been well listened to before. One patient explained some previous experiences of GP appointments:

Because of the nature of how difficult it is to see the same doctor each time, I find it very difficult for them to have the time to listen and really understand what's going on, especially because my needs are sometimes quite complex. So yeah, I do find it very difficult. (Patient 23, Feasibility Study)

This patient went on to explain that the GP in the COAC appointment had made it clear they read the form and then really listened to her:

it enabled me to feel like I can talk about things and say exactly what was going on, which just opens up the patient/doctor relationship more. That means that I'm more likely to see the doctor again next time if I have concerns, which I think, certainly from my point of view, makes it a lot safer. (Patient 23, Feasibility Study)

As a result of the GP making it clear they read the form, this patient felt listened to and taken seriously and more confident in seeking healthcare in the future.

4 Discussion

4.1 Main findings

This paper reports on the person-based development of a pre-consultation form and on a realist evaluation of this which was embedded within a feasibility study. The person-based development was highly successful. Numerous improvements were made and GPs and patients agreed the final version was much improved on the initial version. In the feasibility study the pre-consultation questionnaire was tested in a single intervention with a summary report. Through the embedded realist evaluation, we found that these were useful for different types of patient. The pre-consultation form is most useful for patients with complex problems, mental health issues, health concerns, a concern they find it difficult to voice, or who find consultations nerve-racking. It was also useful for patients who sometimes feel that the GP doesn't listen to them or understand their problems. It was less useful when patients who completed it had a quick problem, or when they had underlying chronic problems that were unrelated to their consultation.

We identified six possible outcomes of the pre-consultation form which is captured in our finalised programme theory. The previous programme theory (Figure 3) had included reduction in re-consultation rates as an outcome. However, interviews did not show any evidence that this is a likely outcome. The two outcomes with the most qualitative evidence were: 1) issues discussed which might not have been discussed otherwise and 2) a wider range of tailored support offered to patients. There was also evidence of more time efficient consultations, greater patient satisfaction, increased confidence in seeking healthcare and improved well-being health knowledge and support.

4.2 Strengths and limitations

The person-based approach was an effective method of developing the pre-consultation and consultation summary reports. The PPI group were actively engaged and an important part of designing of the intervention. Practices were effectively re-engaged following the COVID-19 related study pause. The REDCap system used for data collection enabled patient-reported data to be collected accurately with no data entry error. An effective collaboration was developed with the GP Federation for BNSSG CCG (One Care) who worked were able to publish the required EMIS resources to the GP practices. Pre-consultation form completion rates increased from 15%–17% in the first two practices to 50% in the final practice.

The realist evaluation is a well-established theory-based approach for making sense of why, when and for whom context

sensitive outcomes occur in complex interventions, such as the pre-consultation form. We had a wealth of data across 63 interviews and used rigorous methods to analyse this data within a 3-person team. We were unable to interview non-responders, so we do not know about the type of person who did not respond to the questionnaire. Furthermore, the realist evaluation was restricted to people who agreed to an interview, thereby forming a self-selecting group who either displayed a level of altruism or were more likely to be engaged by the intervention, and this may have affected the findings.

4.3 Comparison with literature

There are varied levels of existing quantitative evidence for the outcomes, mechanisms and contexts identified in our programme theory. Our interviews identified that the questionnaires uncovered issues that might not otherwise have been discussed. This included when the patient had a sensitive problem that they might be reluctant to disclose. Previous studies have found that electronic questionnaires are well suited to exploring issues of a complex, personal or sensitive nature, as patients may be more willing raise sensitive problems on a questionnaire than verbally43-45. A 2020 systematic review showed that use of pre-consultation electronic forms helped patient to disclose more psychosocial and quality of life issues⁴⁶. The form also helped to pick up on low mood and anxiety. Another 2020 systematic review found that primary care patients were often reluctant to disclose emotional concerns that included sub-clinical low mood, stress and/or anxiety, and low mood, stress and/or anxiety attributable to difficult life circumstances⁴⁷.

Making consultations more efficient was an important outcome perceived by patients and GPs. Sometimes this involved speeding the consultation up, and sometime using time to better effect. GP consultations are in the UK are 10.9 minutes long on average. GPs in England spend longer with patients who have more conditions, but, at all multimorbidity levels, those in deprived areas have less time per GP consultation⁴⁸. This intervention could help free up GP time to use with the patients who most need it.

The intervention also seemed to improve patient satisfaction. Good physician-patient communication is central to good patient experience, and a major driver of overall patient assessments of primary care49. The pre-consultation form helped communication on a number of levels: by helping the patient raise problems, helping the GP prepare and focus on what the patient needs and helping the patient feel that they were listened to and taken seriously. Patients were more likely to raise problems about low mood and health concerns if invited to on the form. Previous research has also shown that patients with mental health needs may fail to access support for these needs if they believe they fall outside the scope of primary care in the absence of any physical symptoms^{50,51}. A specific invitation to share their needs may help with this barrier to access⁴⁵. Previous cohort studies have also shown high patient satisfaction following completion of a pre-consultation questionnaire, however studies with a control are have normally shown no difference in patient satisfaction⁴⁶.

A number of patients commented that the GP reading their questionnaire, letting them know they had read it and then listening made them feel more listened to and taken seriously than they had done before. Previous qualitative studies have identified feeling "listened to" as hugely important element of the consultation⁵². Jagosh *et al.* identified that listening was (a) an essential component of clinical data gathering and diagnosis; (b) a healing and therapeutic agent; and (c) as a means of fostering and strengthening the doctor-patient relationship¹³. In our programme theory, we similarly identify that listening is a mechanism for improved patient satisfaction, increased patient confidence in seeking healthcare, and improved wellbeing, as listening is therapeutic in itself.

Research shows that, alongside the switch to telephone and video consultation in March 2020, there was a concurrent reduction in diagnoses in primary care⁵³ and a reduction in the number of routine monitoring tasks by GPs, including health promotion⁵⁴. Our study provides qualitative evidence that the COAC pre-consultation form might help increase diagnoses and monitoring in telephone consultations by helping patients disclose more concerns and focussing GPs on what matters to the patient.

Previous studies have found that electronic triage forms are commonly used for medication-related queries, administrative requests and reporting specific symptoms⁵⁵. The COAC pre-consultation form was, in contrast, used for a range of different types of problems. While some people used them for very simple problems, other patients included a range of problems, non-specific symptoms and health concerns or low mood.

4.4 Conclusions

The pre-consultation form was developed and tested using rigorous methods and has been demonstrated to be valuable for both patients and GPs. It is most useful for patients with complex problems, mental health issues, health concerns, a concern they find it difficult to voice, or who find consultations nerve-racking. For these patients, it can reveal issues which might not have been discussed otherwise and lead to a wider range of tailored support offered to patients. It may also make consultations more time efficient and lead to greater patient satisfaction and well-being. However, the administrators implementing the process required too much support from the study research teams for it to be practical to roll-out using the current technological platform. This paper provides information on how to develop such a technology and describes why it works, for what patients under what circumstances for the benefit of future developers of similar technologies.

Data availability

Underlying data

Researchers can apply for this data via a form on the repository:

University of Bristol: COAC Study Qualitative Dataset, https:// doi.org/10.5523/bris.11jvagu1sigje2duqj3ube527y (restricted access)⁵⁶. This project contains the qualitative data transcripts for the COAC Feasibility Study, where participants agreed that these could be shared with bona fide researchers outside the Bristol research team. Information about each transcript is listed below, as follows:

Transcript ID: The name of the transcript in the folder. The name consists of:

- * a participant identifier
- * the type of participant (patient, clinician or administrator)

* the site (1 to 4 – this was not reported in the paper for reasons of anonymity)

* The date of the interview

Participant identifier used in papers: This is the identifier used in this paper.

The folder also contains the consent form. All patients in this study consented to point 7 in this form: "I understand that after the study my anonymised data will be made available to bona fide researchers for future research studies, and it will not be possible to identify me from these data. If I agree to this, my data will be held for twenty years."

This dataset has an access level *Restricted*, which means it is not available via direct download but must be requested. Research participants did not give explicit consent to share this data as open data but agreed that it should be made available to approved bona fide researchers only, after their host institution has signed a Data Access Agreement. In order to request access to this data please complete the data request form available from the link above. We will consider any application from any organisation where an established research governance process is in place.

Data are available under the terms of a non-Commercial Government Licence for public sector information.

Extended data

University of Bristol: COAC Study Extended Dataset, https://doi. org/10.5523/bris.386dsq2e4iii225ms7du8pd5jq⁵⁷

This project contains the following extended data:

- COAC-pre-consultationForm.doc
 - This file contains screenshots of the pre-consultation form which patients responded to in the COAC Study.
- COACStudy-pre-consultationform-TableOfChanges.doc This file contains a detailed table of changes made to the pre-consultation form in the COAC Intervention Study. Patients who are quoted in this table all consented to the first six points in the consent form included in this folder.
- COACStudy-SummaryReport-TableOfChanges.doc

This file contains a detailed table of changes made to the summary report in the COAC Intervention Study. Patients who are quoted in this table all consented to the first six points in the consent form included in this folder.

• COACStudy-TopicGuides.doc

This file contains the interview topics guides for the COAC Study.

- PatientConsent-Interviewsv1.3.doc This is the patient consent form used for the COAC Study
- PatientInfoInterviewStudy2v1.4.doc

This is the patient information leaflet given to patients interviewed for the COAC Study

• COREQ checklist - pre-consultation form

This is a checklist for the COREQ reporting guidelines which demonstrates how they were following in collecting and analysing data about the pre-consultation form

• COREQ checklist – summary report

This is a checklist for the COREQ reporting guidelines which demonstrates how they were following in collecting and analysing data about the summary report

Reporting guidelines

University of Bristol: COREQ checklist⁵⁸ for COAC Study, https://doi.org/10.5523/bris.386dsq2e4iii225ms7du8pd5jq⁵⁷.

The realist evaluation also followed the RAMESES II reporting standards for realist evaluations⁵⁹.

Data are available under the terms of the Creative Commons Attribution 4.0 International license (CC-BY 4.0).

Acknowledgements

We would like to acknowledge the valuable input of the PPI group, including Tom Yardley, Christina Stokes, Anna Ferguson-Montague, Fatima Ahmed, Mary Ellis and the other members who helped to design the intervention and interpret the findings. We would like to thank Grace Mander from One Care for assisting with development of the EMIS searches, Anna Rushowski, Andrew Appleton and Lakhwinder Anota for helping design the administrative process and user-guide for the pre-consultation form, Victoria Wilson for assisting with co-ordination of the PPI meetings, Jude Hancock for helping to review the Table of Changes and Calum Masterson from the CRN for assistance in recruiting practices. Thanks to advisory group members, Pete Bower, Gary Abel, Julia Frost, Louise Ting and Joanne Protheroe for providing advice and support. Finally, we are very grateful to all the recruiting GPs and patient participants whose input was essential but cannot be acknowledged by name for the purposes of maintaining anonymity.

References

- Marvel MK, Epstein RM, Flowers K, *et al.*: Soliciting the patient's agenda: have we improved? *JAMA*. 1999; 281(3): 283–7. PubMed Abstract | Publisher Full Text
- Hay A, Ridd M: Teachers' and Students' Handbook for Consultation Skills Teaching. MB ChBConsultation Skills: UNIVERSITY OF BRISTOL MEDICAL SCHOOL: Centre for Academic Primary Care, 2014. Reference Source
- Salisbury C, Chalder M, Manku-Scott T, et al.: The national evaluation of NHS walk-in centres. University of Bristol, 2002. Reference Source
- Atherton H, Brant H, Ziebland S, et al.: The potential of alternatives to face-to-face consultation in general practice, and the impact on different patient groups: a mixed-methods case study. Southampton (UK), 2018. PubMed Abstract | Publisher Full Text
- Curtis L, Burns A: Unit Costs of Health and Social Care 2016. Personal Social Services Research Unit, 2016. Reference Source
- Salisbury C, Procter S, Stewart K, et al.: The content of general practice consultations: cross-sectional study based on video recordings. Br J Gen Pract. 2013; 63(616): 751–59.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Rosendal M, Carlsen AH, Rask MT, et al.: Symptoms as the main problem in primary care: A cross-sectional study of frequency and characteristics. Scand J Prim Health Care. 2015; 33(2): 91–9.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Corrie C, Finch A: Expert Patients. Reform, 2015. Reference Source
- Jones CH, Neill S, Lakhanpaul M, et al.: The safety netting behaviour of first contact clinicians: a qualitative study. BMC Fam Pract. 2013; 14: 140. PubMed Abstract | Publisher Full Text | Free Full Text
- Greenhalgh J, Dalkin S, Gooding K, et al.: Functionality and feedback: a realist synthesis of the collation, interpretation and utilisation of patientreported outcome measures data to improve patient care. HEALTH SERVICES AND DELIVERY RESEARCH. Southampton (UK), 2017. PubMed Abstract | Publisher Full Text
- Kessels RP: Patients' memory for medical information. J R Soc Med. 2003; 96(5): 219–22.
- PubMed Abstract | Free Full Text
- McWhinney I: A Textbook of Family Medicine. New York: Oxford University Press, 1997.
 Reference Source
- Jagosh J, Donald Boudreau J, Steinert Y, et al.: The importance of physician listening from the patients' perspective: enhancing diagnosis, healing, and the doctor-patient relationship. Patient Educ Couns. 2011; 85(3): 369–74. PubMed Abstract | Publisher Full Text
- Boudreau JD, Cassell E, Fuks A: Preparing medical students to become attentive listeners. Med Teach. 2009; 31(1): 22–9.
 PubMed Abstract | Publisher Full Text
- Rabinowitz I, Luzzati R, Tamir A, et al.: Length of patient's monologue, rate of completion, and relation to other components of the clinical encounter: observational intervention study in primary care. BMJ. 2004; 328(7438): 501-2.
- PubMed Abstract | Publisher Full Text | Free Full Text 5. Robertson K: Active listening: More than just paying attention. Aust Fam
- Robertson K: Active listening: More than just paying attention. Aust Fam Physician. 2005; 34(12): 1053–55.
 PubMed Abstract
- Jones V: Physician Goes First: A way not to interrupt patients. In: Pho K, ed. KevinMDcom, 2014.
 Reference Source
- 18. CCG CWS: Top tip for people with mental health difficulties to get the best out of your GP & practice. 2017.
- Chen J, Ou L, Hollis SJ: A systematic review of the impact of routine collection of patient reported outcome measures on patients, providers and health organisations in an oncologic setting. *BMC Health Serv Res.* 2013; 13: 211.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Snyder CF, Herman JM, White SM, et al.: When using patient-reported outcomes in clinical practice, the measure matters: a randomized controlled trial. J Oncol Pract. 2014; 10(5): e299–306.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Velikova G, Booth L, Smith AB, et al.: Measuring quality of life in routine oncology practice improves communication and patient well-being: a randomized controlled trial. J Clin Oncol. 2004; 22(4): 714–24. PubMed Abstract | Publisher Full Text
- Hagelin CL, Wengström Y, Tishelman C, et al.: Nurses' experiences of clinical use of a quality of life instrument in palliative care. Contemp Nurse. 2007; 27(1): 29–38.
 PubMed Abstract

- Fayers PM: Evaluating the effectiveness of using PROs in clinical practice: a role for cluster-randomised trials. *Qual Life Res.* 2008; 17(10): 1315–21. PubMed Abstract | Publisher Full Text
- 24. Elmore N: Recruiting patients to research? Our top tips on working that waiting room. Cambridge Centre for Health Services Research, 2013.
- Murphy M, Hollinghurst S, Cowlishaw S, et al.: Psychometric Testing of the Primary Care Outcomes Questionnaire. British Journal of General Practice. Pending publication, 2018.
- Honeyman M, Dunn P, McKenna H: A digital NHS? An introduction to the digital agenda and plans for implementation. 2016. Reference Source
- 27. NHS England: The NHS Long Term Plan. 2019. Reference Source
- Craig P, Dieppe P, Macintyre S, et al.: Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008; 337: a1655.
 - PubMed Abstract | Publisher Full Text | Free Full Text
- Yardley L, Morrison L, Bradbury K, et al.: The person-based approach to intervention development: application to digital health-related behavior change interventions. J Med Internet Res. 2015; 17(1): e30. PubMed Abstract | Publisher Full Text | Free Full Text
- Murphy M, Hollinghurst S, Salisbury C: Qualitative assessment of the primary care outcomes questionnaire: a cognitive interview study. BMC Health Serv Res. 2018; 18(1): 79.
 PubMed Abstract | Publisher Full Text | Free Full Text
- 31. Vanderbilt University: **Redcap website.** 2017. **Reference Source**
- Hayes H, Buckland S, Involve TM: Briefing notes for researchers: public involvement in NHS public health and social care research. NIHR, 2012. Reference Source
- Salisbury C, Man MS, Bower P, et al.: Management of multimorbidity using a patient-centred care model: a pragmatic cluster-randomised trial of the 3D approach. Lancet. 2018; 392(10141): 41–50.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Murphy M, Scott A, Wong G, et al.: The Consultation Open and Close Study: A Feasibility Study of a complex intervention. pending submission to NIHR Open planned 2022.
- Ford I, Norrie J: Pragmatic Trials. N Engl J Med. 2016; 375(5): 454–63.
 PubMed Abstract | Publisher Full Text
- Campbell R, Starkey F, Holliday J, et al.: An informal school-based peer-led intervention for smoking prevention in adolescence (ASSIST): a cluster randomised trial. Lancet. 2008; 371(9624): 1595-602.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Campbell JL, Fletcher E, Britten N, *et al.*: Telephone triage for management of same-day consultation requests in general practice (the ESTEEM trial): a cluster-randomised controlled trial and cost-consequence analysis. *Lancet.* 2014; 384(9957): 1859–68.
 PubMed Abstract | Publisher Full Text
- Dey I: Grounding grounded theory: Guidelines for qualitative inquiry. San Diego, CA: Academic Press. 1999).
 Reference Source
- 39. Pawson R, Tilley N: Realistic Evaluation. London: SAGE Publications. 1997. Reference Source
- 40. Authority NHR: South West Frenchay Research Ethis Committee Meeting Dates. 2021. Reference Source
- Murphy M, Salisbury C, Palmer T, et al.: Using health information systems to address patients concerns in general practice: the COAC Intervention development and feasibility study. | Research Square. Research Square (Pre-print). 2021.
 Publisher Full Text
- Murphy M, Hollinghurst S, Cowlishaw S, et al.: Primary Care Outcomes Questionnaire: psychometric testing of a new instrument. Br J Gen Pract. 2018; 68(671): e433–e40.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Cooley PC, Rogers SM, Turner CF, et al.: Using Touch Screen Audio-CASI to Obtain Data on Sensitive Topics. Comput Human Behav. 2001; 17(3): 285–93. PubMed Abstract | Publisher Full Text | Free Full Text
- Goodhart IM, Andrzejowski JC, Jones GL, et al.: Patient-completed, preoperative web-based anaesthetic assessment questionnaire (electronic Personal Assessment Questionnaire PreOperative): Development and validation. Eur J Anaesthesiol. 2017; 34(4): 221–28. PubMed Abstract | Publisher Full Text
- Gerbert B, Bronstone A, Pantilat S, et al.: When asked, patients tell: disclosure of sensitive health-risk behaviors. Med Care. 1999; 37(1): 104–11.
 PubMed Abstract | Publisher Full Text
- 46. Wickramasekera N, Taylor SK, Lumley E, et al.: Can electronic assessment

tools improve the process of shared decision-making? A systematic review. Health Inf Manag. 2020; 1833358320954385. PubMed Abstract | Publisher Full Text

- Parker D, Byng R, Dickens C, et al.: Barriers and facilitators to GP-patient communication about emotional concerns in UK primary care: a systematic review. Fam Pract. 2020; 37(4): 434-44.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Gopfert A, Deeny SR, Fisher R, et al.: Primary care consultation length by deprivation and multimorbidity in England: an observational study using electronic patient records. Br J Gen Pract. 2021; 71(704): e185-e92.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Paddison CA, Abel GA, Roland MO, et al.: Drivers of overall satisfaction with primary care: evidence from the English General Practice Patient Survey. Health Expect. 2015; 18(5): 1081–92.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Kravitz RL, Paterniti DA, Epstein RM, et al.: Relational barriers to depression help-seeking in primary care. Patient Educ Couns. 2011; 82(2): 207–13.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Biddle L, Donovan J, Sharp D, et al.: Explaining non-help-seeking amongst young adults with mental distress: a dynamic interpretive model of illness behaviour. Sociol Health Illn. 2007; 29(7): 983–1002.
 PubMed Abstract | Publisher Full Text
- 52. Murphy M, Salisbury C: Relational continuity and patients' perception of GP

trust and respect: a qualitative study. Br J Gen Pract. 2020; **70**(698): e676–e83. PubMed Abstract | Publisher Full Text | Free Full Text

- Williams R, Jenkins DA, Ashcroft DM, et al.: Diagnosis of physical and mental health conditions in primary care during the COVID-19 pandemic: a retrospective cohort study. Lancet Public Health. 2020; 5(10): e543–e50. PubMed Abstract | Publisher Full Text | Free Full Text
- Murphy M, Scott LJ, Salisbury C, et al.: Implementation of remote consulting in UK primary care following the COVID-19 pandemic: a mixed-methods longitudinal study. Br J Gen Pract. 2021; 71(704): e166–e77.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Eccles A, Hopper M, Turk A, et al.: Patient use of an online triage platform: a mixed-methods retrospective exploration in UK primary care. Br J Gen Pract. 2019; 69(682): e336-e44.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Murphy M, Scott A, Salisbury C, et al.: COAC Study Qualitative Dataset. 2022
- 57. Murphy M, Salisbury C, Macleod J: **COAC Study Extended Dataset.** 2022.
- Tong A, Sainsbury P, Craig J: Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. Int J Qual Health Care. 2007; 19(6): 349–57.
 PubMed Abstract | Publisher Full Text
- Wong G, Westhorp G, Manzano A, et al.: RAMESES II reporting standards for realist evaluations. BMC Med. 2016; 14(1): 96.
 PubMed Abstract | Publisher Full Text | Free Full Text

Open Peer Review

Current Peer Review Status:

Version 2

Reviewer Report 19 August 2022

https://doi.org/10.3310/nihropenres.14409.r28730

© **2022 Eccles A.** This is an open access peer review report distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.



Abi Eccles 匝

Unit of Academic Primary Care, Warwick Medical School, University of Warwick, Coventry, UK

Authors have done a great job at responding to my comments and making relevant changes. The examples of patients feeling they can discuss 'hidden' issues are particularly useful demonstrating an important advantage of this tool.

Thank you for citing my paper on online triage. However, I still think readers would benefit from further explanation about how this tool differs to online triage. You have explained that the tool could be used to a range of concerns, but this does not distinguish it from online triage where patients are not limited in what they can write. Some discussion about how this tool is more sophisticated, perhaps how was developed, how it actively encourages patients to highlight a broader range of problems, or content is directly used using the consultation would be useful.

Competing Interests: No competing interests were disclosed.

Reviewer Expertise: Qualitative health researcher working on a range of digital access projects.

I confirm that I have read this submission and believe that I have an appropriate level of expertise to confirm that it is of an acceptable scientific standard.

Version 1

Reviewer Report 09 May 2022

https://doi.org/10.3310/nihropenres.14365.r28574

© **2022 Eccles A.** This is an open access peer review report distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

? 🛛 Abi Eccles 匝

Unit of Academic Primary Care, Warwick Medical School, University of Warwick, Coventry, UK

Thank you for inviting me to review this paper, which was very interesting to read. This is very comprehensive and rigorous work evaluating the development and feasibility of a consultation tool. The problem and aims were laid out well. The lay summary is very clear and engaging. There was excellent detail and background describing the CMOC process.

There are some areas that would benefit from further refinement. It would be useful to have some more detail regarding the methods. So, how participants were selected for interviews if there was a sampling strategy. I was also interested to know what happened to the control group. I appreciate that is not the nature of this evaluation, but perhaps a brief note or reference to where it would be reported might be useful

It is a little unclear whether the REDCap questionnaire was collecting baseline data or the pre consultation tool or both. One of the flow diagrams just says baseline. It might be helpful to clarify this and add a note that it is the pre consultation tool (as well as the baseline data collection) in any diagrams.

A definition of BNF might be useful for those unfamiliar.

Some areas that could do we more consideration and discussion include:

- Patients' hidden concerns or worries I was not convinced that the tool uncovered these if it is just based on a judgement GPs made. Was there any evidence from what patients said?
- Those patients who opted not to complete the form. What percentage didn't? Was there any indication why it was not appealing to those individuals? What can be done to engage them? Also, has there been any further consideration to transfer to non-digital (as highlighted by the PPI group).

It would be interesting to know more about how long the form took to complete for patients and whether the pre consultation review added much admin time to GP's workload.

This pre-consultation questionnaire has some similarities to the online triage initiatives (AskMyGP econsults). I think it would be useful to refer to these and point out explicitly how this tool differs. There are some similarities in literature describing the impacts of these online triage tools for patients and GPs too, which would be useful to reference.

There was a few typos:

- Pg 3 apostrophes missing from "patients reason" (p.3) "patients interests" (p.4).
- I suggest using term co-investigators, rather than co-applicants (as funded) (p.7,8).
- P.8 "assigned a code to it" is a little unclear. Do you mean assigned a code to each change suggested? Could perhaps clarify.
- Inconsistency with randomisation/randomization (p.9).
- Word missing: "Researcher then took informed ..." (p.9).

- Letter missing: Figure 14: 'Feeing' instead of 'feeling'.
- Letter missing: 'Though' instead of 'thought' (p.17+24+24+26).

Is the work clearly and accurately presented and does it cite the current literature? Partly

Is the study design appropriate and is the work technically sound? $\ensuremath{\mathsf{Yes}}$

Are sufficient details of methods and analysis provided to allow replication by others? Partly

If applicable, is the statistical analysis and its interpretation appropriate? Not applicable

Are all the source data underlying the results available to ensure full reproducibility? Partly

Are the conclusions drawn adequately supported by the results? Yes

Competing Interests: No competing interests were disclosed.

Reviewer Expertise: Qualitative health researcher working on a range of digital access projects.

I confirm that I have read this submission and believe that I have an appropriate level of expertise to confirm that it is of an acceptable scientific standard, however I have significant reservations, as outlined above.

Author Response 14 May 2022

Mairead Murphy, University of Bristol, Bristol, UK

Dear Abi, Thanks for this useful review. We will certainly make some changes based on this and submit another version. Mairead

Competing Interests: No competing interests were disclosed.

Reviewer Report 20 April 2022

https://doi.org/10.3310/nihropenres.14365.r28554

© **2022 Teeling S.** This is an open access peer review report distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.



Seán Paul Teeling 问

¹ Health Sciences Centre, School of Nursing Midwifery and Health Systems, University College Dublin (UCD), Dublin, Ireland

² Centre for Person-Centred Practice research, Queen Margaret University, Edinburgh, UK

Thank you for the opportunity to review this excellent paper. The paper addresses a key issue in health services internationally - the lack of time available to busy clinicians to spend the time they would like to spend with their patients, leading to a consultation that can be about the condition rather than the person. The paper clearly details the iterative refinement of the consultation form through engagement with both patients and the GP practice teams so that the 'customer voice' is captured throughout. It is clear that this led to the development of a more person-centred consultation process that works for both patients and the practice staff. The paper illustrates important contextual factors and mechanisms within the process of GP patient consults that can be applicable to those seeking to improve both patient and staff experiences of care.

Is the work clearly and accurately presented and does it cite the current literature? $\ensuremath{\mathsf{Yes}}$

Is the study design appropriate and is the work technically sound?

Yes

Are sufficient details of methods and analysis provided to allow replication by others? $\ensuremath{\mathsf{Yes}}$

If applicable, is the statistical analysis and its interpretation appropriate? Not applicable

Are all the source data underlying the results available to ensure full reproducibility? $\ensuremath{\mathsf{Yes}}$

Are the conclusions drawn adequately supported by the results?

Yes

Competing Interests: No competing interests were disclosed.

Reviewer Expertise: Person-centred care, Person-centredness, Person-centred Cultures, Process Improvement, Quality Improvement, Realist Evaluation.

I confirm that I have read this submission and believe that I have an appropriate level of expertise to confirm that it is of an acceptable scientific standard.

Author Response 29 Apr 2022

Mairead Murphy, University of Bristol, Bristol, UK

Thank you Sean for your kind words and for this helpful review. Mairead

Competing Interests: No competing interests were disclosed.